Copyright & Permissions

Copyright © 1998, 1999, 2000, 2001, 2002, 2003, 2004, 2005, 2006, 2007, 2008, 2009, 2010, 2011, 2012, 2013, 2014, 2015, 2016, 2017, 2018, 2019, 2020, 2021, 2022, 2023 by the American Economic Association.

Permission to make digital or hard copies of part or all of American Economic Association publications for personal or classroom use is granted without fee provided that copies are not distributed for profit or direct commercial advantage and that copies show this notice on the first page or initial screen of a display along with the full citation, including the name of the author. Copyrights for components of this work owned by others than AEA must be honored. Abstracting with credit is permitted.

The author has the right to republish, post on servers, redistribute to lists and use any component of this work in other works. For others to do so requires prior specific permission and/or a fee. Permissions may be requested from the American Economic Association Administrative Office by going to the Contact Us form and choosing "Copyright/Permissions Request" from the menu.

Copyright © 2023 AEA

The Demand for Energy-Using Assets among the World's Rising Middle Classes[†]

By PAUL J. GERTLER, ORIE SHELEF, CATHERINE D. WOLFRAM, AND ALAN FUCHS*

We study household decisions to acquire energy-using assets in the presence of rising incomes. We develop a theoretical framework to characterize the effect of income growth on asset purchases when consumers face credit constraints. We use large and plausibly exogenous shocks to household income generated by the conditional-cash-transfer program in Mexico, Oportunidades, to show that asset acquisition is nonlinear, depends, as predicted in the presence of credit constraints, on the pace of income growth, and both effects are economically large among beneficiaries. Our results may help explain important worldwide trends in the relationship between energy use and income growth. (JEL D12, I32, I38, O12, O13, Q47)

Energy is a fundamental input to modern life. Without access to commercial energy sources, such as gasoline, natural gas, and electricity, people could not drive vehicles, refrigerate food and medicine, air condition buildings, watch television, easily operate farming equipment, or participate in many other aspects of modern life. Despite this, an estimated 1.3 billion people live without electricity in their homes, and even among those who have access, many do not own basic assets such as refrigerators, motorized transport, or washing machines. In fact, Table 1 demonstrates the low penetration of several key energy-using assets for over 4 billion people living in the developing world, especially when compared to high-income countries such as the United States. However, this situation is rapidly changing as incomes rise from economic growth and as massive poverty alleviation programs continue to expand.

[†]Go to http://dx.doi.org/10.1257/aer.20131455 to visit the article page for additional materials and author disclosure statement(s).

^{*}Gertler: Haas School of Business, University of California at Berkeley, Berkeley, CA 94720, and NBER (e-mail: gertler@haas.berkeley.edu); Shelef: Stanford Institute for Economic Policy Research, Stanford University, 366 Galvez Street, Stanford, CA 94305 (e-mail: oshelef@stanford.edu); Wolfram: Haas School of Business, University of California at Berkeley, Berkeley, CA 94720, and NBER (e-mail: cwolfram@berkeley.edu); Fuchs: The World Bank, Poverty and Equity Global Practice, 1818 H Street, NW I 4-405, Washington, DC 20433 (e-mail: afuchs@worldbank.org). We are grateful to Arthur Van Benthem, Lucas Davis, Chris Knittel, Jakob Svensson, and Deng Yongheng as well as seminar participants at Iowa State, MIT, Stanford, University of Toronto, Tufts, Washington University, and attendees of the Energy Institute at Haas Summer Camp and the Stockholm Climate-Economy Conference for valuable comments. We previously circulated this paper with the title "How Pro-Poor Growth Affects the Demand for Energy." We declare that we have no relevant or material financial interests that relate to the research described in this paper. The findings, interpretations, and conclusions expressed in this work do not necessarily reflect the views of the World Bank, its Board of Executive Directors, or the governments they represent. The World Bank does not guarantee the accuracy of the data included in this work.

	Electricity access (percent of population)	Refrigerators (share of households)	Cars (per 1,000 people)	Population (millions)
Brazil	98.7	0.93	209	197
China	99.7	0.69	58	1,344
India	75.0	0.13	18	1,221
Indonesia	73.0	0.17	60	245
Mexico	97.9	0.83	275	119
Sub-Saharan Africa	32.5	0.11	28	886
Total	70.8	0.38	53	4,012
United States	100.0	0.99	797	312

Notes: Population numbers are from 2011. Data on electricity access and cars are from 2008–2010, except Mexico electricity access is from 2012. Refrigerator shares come from a variety of country-specific nationally representative surveys for the following years: Brazil (2009); China (2010); India (2007–2008); Indonesia (2004); Mexico (2008); sub-Saharan Africa: aggregated country-level surveys (2006); United States (2011).

Sources: National Bureau of Statistics (2011); United States Census Bureau (2013); Wolfram, Shelef, and Gertler (2012); World Bank (2011).

We analyze household decisions to acquire energy-using assets, focusing on the role of rising incomes in the developing world where the vast majority of the growth in energy use is expected (Energy Information Administration (EIA) 2013). Importantly, and in contrast to previous literature, we allow for the presence of credit constraints, which we show have significant implications for the timing of asset acquisitions and, consequently, for the demand for energy. This is important as credit is severely constrained throughout much of the developing world, especially for the poor and near poor (Karlan and Morduch 2010).¹ In rural Mexico, the site of our empirical application, credit opportunities are very limited as only 1 percent of communities have a formal credit institution and only 3 percent of households report having active loans, almost all of which are small informal loans from friends and relatives (Gertler, Martinez, and Rubio-Codina 2012).

We are interested in acquisition of energy-using assets for two reasons. First, they are important drivers of health and human development (World Bank 2008). Air conditioning lowers heat-related mortality (Barreca et al. forthcoming) and refrigeration improves child health outcomes (Wolfe and Behrman 1982). There are also established causal linkages between access to electricity and female labor force participation (Dinkelman 2011), housing values, and the UN Human Development Index (Lipscomb, Mobarak, and Barham 2013). While there is some work on the acquisition of energy-using assets in the developed world,² ours is the first to analyze the acquisition of these goods in the developing world.³

¹See also http://datatopics.worldbank.org/financialinclusion/.

²See, for example, recent contributions on heating systems by Davis and Kilian (2011) and automobiles by Busse, Knittel, and Zettelmeyer (2013).

³ In some countries, there are large upfront prices for an electricity connection. In Kenya, for example, the cost of connecting to the electricity grid is around \$400, about one-third of the average annual income in the country. By contrast, other countries, such as South Africa, have subsidized the cost of the connection entirely. Connection costs in Mexico are on the low end of this range, though not zero. Our model highlights the potential role of income growth and credit markets in increasing connections, another asset households acquire, in places like Kenya.

JUNE 2016

Second, the growth in energy demand driven by rising energy use among first-time purchasers of household appliances and other energy-using assets is likely to have important implications for macro-level trends in energy use. In terms of scale, if, for example, one-half of the households in India who do not own refrigerators were to buy one, annual *nationwide* electricity demand across all sectors—residential, commercial, and industrial—would rise by over 10 percent.⁴ Rapid first-time asset acquisition is already taking place in some parts of the world. For instance, vehicle ownership in urban China has risen at almost 40 percent *per year* between 2000 and 2010, helping fuel China's rapid growth in oil consumption (National Bureau of Statistics 2001, 2011).

Understanding the likely growth in demand for energy is critical for several reasons. First, investments in energy infrastructure require long lead times, and most governments and energy companies base their investment decisions on demand projections. Incorrect forecasts can lead to local energy shortages that affect both productivity and welfare (Allcott, Collard-Wexler, and O'Connell 2016). On a global scale, faster than anticipated growth in energy demand can lead to significant increases in energy prices. Second, energy use is a key contributor to climate change as energy-related emissions account for three-quarters of worldwide anthropogenic greenhouse gas emissions.⁵ Forecasting the likely path of greenhouse gas emissions is essential to understanding the range of possible effects of climate change. Furthermore, expected country-level emissions are critical inputs to any plans to mitigate climate change, including an international climate agreement.

We demonstrate elsewhere that existing energy forecasts do not appear to account sufficiently for the nonlinear relationship between energy and income (Wolfram, Shelef, and Gertler 2012). This paper begins by documenting motivating evidence of a nonlinear relationship between income and both asset ownership and energy use at the individual, municipality, and country levels. Since households do not consume energy directly but rather through appliances, this suggests that there is an empirical regularity to be explained with micro evidence on appliance purchases. We next develop a simple theoretical model that suggests a nonlinear Engel curve, meaning that as income goes up from initially very low levels, credit-constrained households do not become more likely to purchase an energy-using asset. Above a certain threshold, however, increases in income are much more likely to lead to asset purchases, suggesting a nonlinear relationship between income and asset acquisition. Our theoretical model also predicts that the speed at which credit-constrained households' incomes grow will affect their asset acquisitions. The model further predicts a positive interaction between the rate of income growth and income levels, so the impact of rapid income growth is accentuated at higher income levels.

⁴In 2007, 86.5 percent of Indian households did not own refrigerators. Extrapolating growth in acquisitions, by 2011, an estimated 84 percent of households did not own refrigerators (Wolfram, Shelef, and Gertler 2012). Population in 2011 was 1.210 billion. Assuming 5 people per household, 205 million households lack refrigerators. Assuming refrigerators use 1,000 kilowatt hours per year (though new models can be more efficient, this would be a relatively efficient used refrigerator) acquisition by one-half of the households would be 101,648 gigawatt hours per year, which is 10 percent of generation of electricity of 985,443 gigawatt hours in 2011 (EIA 2011). Using generation as the denominator understates the potential impact since line losses mean that more electricity needs to be generated than consumed.

⁵See, for example, http://www.epa.gov/climatechange/science/indicators/ghg/global-ghg-emissions.html.

We next use large and plausibly exogenous shocks to household income generated by the conditional cash transfer program in Mexico, Oportunidades, to empirically show that the nonlinear relationship between income and asset acquisition is important among low-income Mexican households. We also find strong empirical support for both of the predictions on the growth rate of income, providing further validation for the importance of credit constraints and the implied nonlinear relationship between income growth and asset acquisition.

The next section describes motivating evidence on the nonlinear relationship between income and both asset ownership and energy use drawn from several different locations and levels of aggregation. Section II presents a simple two-period model of asset acquisition in the presence of borrowing constraints and varying rates of income growth. Section III describes the Oportunidades program, which we use to test the predictions of our model. Section IV describes our empirical approach. We present results on asset acquisition by Oportunidades households in Section V. Finally, Section VI concludes.

I. Motivating Facts

A. Income and Asset Ownership

We begin by documenting an S-shaped relationship between income and durable asset ownership.⁶ Figure 1 uses household data from several of the most populous developing countries to plot the share of households that own refrigerators against household expenditures per capita.⁷ The dashed lines show the density of households by expenditure level. The top panels depict regions that have experienced recent income growth among the poor, largely driven by poverty alleviation programs in Mexico and Brazil, and economic growth in urban China. As a result, a substantial share of households in these regions has already moved through the income level associated with the inflection point in refrigerator ownership. The bottom row, however, shows that there are still significant populations poised to buy refrigerators in India, Indonesia, and rural China, which together represent more than 2 billion people.

Figure 2 depicts changes in refrigerator ownership over time showing that the nonlinear relationship is also reflected in adoption associated with income growth. The graph shows changes in the share of Mexican households owning a refrigerator by real household consumption level. The gray line reflects changes between 1992 and 1996. At that point, most of the growth in refrigerator ownership was concentrated

⁶Several existing papers have also noted the S-shaped relationship between income and durable asset purchase including Kopits and Cropper (2005); Dargay, Dermot, and Sommer (2007); and Letschert and McNeil (2007). While the existing literature documents the existence of S-shaped cross-sectional correlations, we provide a model to explain it, and show that the relationship is indeed causal.

⁷This paper focuses on refrigerators in part because they are easy to measure consistently across households. Other assets, such as hot water heaters, may range from an outdoor container angled to catch daytime sun to a commercially powered on-demand or stand-alone unit. Refrigerators also represent a sizable share of residential electricity consumption, totaling one-third of total household electricity consumption in Mexico and China, two countries for which we have data (Zhou et al. 2011 and personal communication; National Bureau of Statistics 2011; Johnson et al. 2009). Finally, as households purchase refrigerators, they become consumers of refrigerated products, and considerable energy is used to keep food chilled while it is manufactured, transported, warehoused, and finally sold retail. Estimates suggest that 15 percent of global energy is devoted to the "cold chain" (Twilley 2014), which also accounts for one-third of global emissions of hydrofluorocarbons (HFCs), a potent greenhouse gas (PR Newswire 2014).



FIGURE 1. REFRIGERATOR OWNERSHIP AND HOUSEHOLD EXPENDITURE LEVEL

Notes: Annual expenditure and income per person calculations divides household expenditure and income by the number of adult equivalents in the household, where each household member under 12 years old is treated as half an adult. Frequency weights are used for all surveys other than China, which do not report weights. Brazil uses annual income per person instead of expenditure.

Sources: Mexico, 2008: Encuesta Nacional de Ingresos y Gastos de los Hogares; Brazil, 2009: National Household Sample Survey PNAD; China, 2002: Chinese Household Income Project; India, 2008: National Sample Survey; Indonesia, 2004: National Socio-Economic Survey (SUSENAS).

in the middle consumption levels, with lower growth for both the wealthiest and the poorest households. The black line depicts changes between 2004 and 2008. By this point, new acquisitions are most prevalent among households with the lowest consumption, which is consistent with a general movement of the population through an S-curve. In other words, in the mid-1990s, incomes among the poorest Mexicans were still too low to support refrigerator acquisitions. But as incomes grew and real refrigerator prices fell, the mass of refrigerator purchasers shifted toward poor households and the plot suggests that Mexico has developed such that only the top half of the S-curve is now relevant.

B. Income and Energy Use

One concern is that a nonlinear relationship between income and the purchase of a single asset may disappear when data are aggregated across the purchase of many durable assets. Specifically, even if there is a nonlinear relationship between refrigerators and income, and even if refrigerators are representative of other energy-consuming assets, it is possible that a series of nonlinear relationships at different inflection points may manifest as a linear relationship when aggregated across assets. This could translate into a linear relationship between income and energy demand.



Starting household expenditure (log 2008 pesos)

FIGURE 2. CHANGE IN REFRIGERATOR OWNERSHIP BY HOUSEHOLD EXPENDITURE LEVEL

Notes: The gray line plots the change in the share of households owning a refrigerator between 1992 and 1996 by 1992 expenditures per adult equivalent. The black line plots the change in that share between 2004 and 2008 by 2004 expenditures per adult equivalent. Both lines reflect smoothed values for a locally weighted LOWESS regression excluding the top 1 percent of households by expenditures per adult equivalent. We form a pseudo panel by calculating, at each percentile of the expenditure distribution, the mean household expenditure and change in refrigerator ownership across the base and end years. Expenditure per adult equivalent is deflated by the urban and rural Mexican Consumer Price Indices, though these may be imprecise in the mid-1990s given the currency crisis and they do not necessarily capture changes in refrigerator prices over the time period. Heterogeneous income growth, such as that received by low-income households between 2004 and 2008 due to Oportunidades, will lead to heterogeneous acquisition. Figure 2 reflects household-level, repeated cross-section data from a nationally representative sample.

Sources: Encuesta Nacional de Ingresos y Gastos de los Hogares 1992, 1996, 2004, and 2008.

Figure 3 provides additional support for a nonlinear relationship between income and energy. Panel A of Figure 3 suggests that overall linearity is not the case among Oportunidades households. It plots the log of energy (gas plus electricity) expenditures against the log of total household consumption in 2007. The figure shows an S-shape with a nontrivial share of households below the total consumption level at which the slope of the relationship increases.⁸

The analysis so far applies only to Oportunidades households, which come from the bottom quintile of the income distribution, only reflects the residential sector, and only represents one year, 2007. Panel B of Figure 3 reflects the income-energy use relationship for Mexico as a whole using municipality-level data for 2000–2010.

⁸ The relationship in panel A of Figure 3 does not control for differences across households. We also regressed the log of energy expenditures on the log of total consumption and fully saturated the model with village fixed effects and household-level controls. For households below the twenty-fifth percentile in total consumption the estimated coefficient on log total consumption was -0.013 (standard error = 0.092) while the estimated coefficients are significantly different from one another at the 2 percent level and are consistent with the first part of an S-shaped relationship between energy use and income: i.e., a flat relationship at low-income levels and positive at higher levels.



FIGURE 3. EXAMPLES OF NONLINEAR RELATIONSHIP BETWEEN INCOME AND ENERGY USE

Notes: In panels A to C, the dark, solid line or points plot the smoothed values from a locally weighted LOWESS regression and the dashed line plots the kernel density estimates of the observations. Panel A reflects household-level data. Panel B reflects municipality-level data. In panel B, the income elasticity is estimated for each municipality as the percent change in energy expenditures between 2005 and 2010 divided by the percent change in income. Panel C reflects country-level data and includes countries that reported data consistently from 1980 to 2010 and excluded 15 major oil producers. Requiring data since 1980 excludes the former Soviet Republic and Eastern bloc countries. Appliance ownership rates are very high in those countries given their incomes and weakly correlated with income. Panel D reflects within country level. Countries are categorized as having pro-poor growth if the decrease in the reported Gini coefficient exceeds the median. We include countries for which both energy use and multiple Gini measures are available. Lines are ordinary least squares (OLS) regressions.

Sources: Panel A, 2007 Oportunidades Evaluation Survey (ENCEL). Panel B, World Bank's Poverty and Gender Unit, Latin America and Caribbean region, which uses inputs from the Household Income and Expenditure Survey (ENIGH) of 1992, 2000, 2005, and 2010, and from the population census of 1990, 2000, 2005, and 2010, and employs the small-area estimation methodology proposed by Elbers, Lanjouw, and Lanjouw (2003). Electricity data are from Instituto Nacional de Estadistica, Geografia e Informatica (INEGI) at: http://www3.inegi.org.mx/sistemas/biinegi/. Panels C and D, World Bank's World Development Indicators series (http://databank.worldbank.org/data/views/variableSelection/selectvariables.aspx?source=world-development-indicators).

Panel B plots the smoothed relationship between each municipality's income elasticity of demand for electricity between 2005 and 2010 against baseline per capita income in 2000. The horizontal axis reflects the log of average per capita municipal income in 2000 and the vertical axis reflects the municipal income elasticity of per capita electricity consumption estimated between 2005 and 2010.⁹

If there is an S-shaped relationship between income and energy use that is fixed over time, then as incomes rise, the income elasticity at a point on the income distribution will equal the slope of the S-curve at that point. The inverse-U relationship between the income elasticity of electricity consumption and the income of the municipality in panel B is consistent with the S-shaped relationship: for municipalities with low average incomes, the income elasticity is low but rises in the middle part of the income distribution before falling again at the high points.

We obtain similar patterns to Mexican municipalities using data across countries. Panel C plots the income elasticity of per capita energy consumption estimated between 1985 and 2010 versus the log of average per capita income in 1981. Again, we see an inverse-U relationship, consistent with the S-curve.¹⁰

To demonstrate that cross-household differences could be driving country-level trends, panel D of Figure 3 plots log per capita energy consumption against log per capita GDP, both demeaned at the country level. We include the same set of countries and use the same data as in panel C for 1980–2010. We separate countries into two equally sized bins according to the change in the reported Gini coefficient.¹¹ Countries with larger reductions in their Gini coefficients, suggesting more of the growth has gone to households on the lower end of the income distribution, will have a larger share of households passing through the first inflection point on the S-curve. Panel D of Figure 3 suggests that increases in energy use for a given increase in per capita GDP have been 50 percent higher in countries with pro-poor growth (black) compared to countries with more regressive growth (gray).

In sum, the patterns depicted in this section are consistent with a nonlinear relationship between income and energy consumption both at the household and aggregate levels and suggest that further investigation of the relationship between appliance purchases, which drive energy consumption, and income is warranted.

II. Theoretical Model

This section presents a simple model that elucidates the relationship between income and the purchase of energy-using assets in the presence of credit constraints. In their influential paper, Dubin and McFadden (1984) emphasize that energy consumption depends not only on the usual utility-maximization problem as a function of income and energy prices, but also on the household's current appliance holdings. A number of subsequent papers have analyzed appliance acquisitions, however, few

⁹We plot the income elasticities against 2000 income instead of 2005 income so that the base year for the elasticity calculation is not the same as the scale for the horizontal axis. If it were the same, quicker mean reversion in income than electricity consumption could generate an inverse-U relationship mechanically.

¹⁰ As in panel B, we plot the elasticities against 1981 income so that the base year for the elasticity calculation is different from the year reflected on the horizontal axis.

¹¹Demeaning the data controls for the level of inequality, level of energy use, and other fixed country-specific attributes. Results are very similar if we use alternate definitions of pro-poor growth in place of the change in the Gini coefficient, though Gini is reported most consistently.

JUNE 2016

researchers have analyzed the intertemporal dynamics that may influence these decisions. In fact, most researchers make assumptions that preclude intertemporal considerations, such as perfectly efficient capital markets.¹² While such assumptions may or may not be appropriate in the developed world, it is clear that capital constraints are significant among the poor in the developing world.¹³ In Section IIA we lay out notation and assumptions. Many of the assumptions are for expositional clarity and are not necessary to generate the model predictions. In online Appendix A, we relax several assumptions and demonstrate the results hold more generally. In Section IIB we generate a series of results that allow us to derive the empirically testable predictions in Section IIC.

A. Model Setup

We model decisions by an individual *i* who consumes two goods: a continuous nondurable good that gives per-period utility $u(\cdot)$, and a lumpy durable asset that gives static per-period utility *R* if owned. We normalize the price of the nondurable to 1, and denote the price of the durable as *P*.

Consider a simple two-period model without interest or discounting. We denote consumer *i*'s period 1 income as Y_1 and period 2 income as Y_2 . Define $\overline{Y} = (Y_1 + Y_2)/2$ the average per period income and $a = Y_2/\overline{Y}$ a measure of income growth. We assume that income is weakly increasing so that $a \in [1,2]^{14}$ and that Y and R are uncorrelated and that the consumer has no uncertainty over Y_2 . For expositional clarity, we assume that the price of the good, P, is high enough relative to income levels and growth that any individual who buys in period 2 optimally saves a bit in period 1: $P > 2\overline{Y}(a - 1)$. That is, the difference in income between the two periods is less than P. We make two assumptions on the shape of the consumer's utility function: decreasing marginal utility, i.e., that $u(\cdot)$ has $u'(\cdot) > 0$, $u''(\cdot) < 0$, and an additional assumption on the utility function that is implied by either decreasing risk aversion, prudence, or precautionary savings. See online Appendix A for further details.

B. Results

As a benchmark, absent credit constraints the purchase of the asset does not depend on the time path of income. The household either buys in period 1, or not. Without credit constraints and if not purchasing, the optimal consumption of the

¹²For example, Dubin and McFadden (1984) and, more recently, Bento et al. (2009), assume a perfectly competitive rental market for durables. This could exist in the presence of efficient capital markets and an efficient resale market. In recent work, Rapson (2014) and Schiraldi (2011) model dynamic considerations focusing on, respectively, consumer expectations about future energy (i.e., usage) prices and heterogeneous consumer transaction costs. No papers, of which we are aware, explicitly model credit constraints or analyze durable good acquisition in the developing world. ¹³Liquidity constraints and poverty has been explored in Banerjee and Newman (1993); Aghion and Bolton

¹⁵Liquidity constraints and poverty has been explored in Banerjee and Newman (1993); Aghion and Bolton (1997); Lindh and Ohlsson (1998); Lloyd-Ellis and Bernhardt (2000); Banerjee (2004); and de Mel, McKenzie, and Woodruff (2008) among others and are surveyed in Karlan and Morduch (2010). There are also studies looking at the novel institutions developed to partially overcome credit constraints including: ROSCAs (Besley, Coate, and Loury 1993) and microfinance (Hossian 1988). None of these intuitions address household durables, either in intent or scale.

¹⁴Without this assumption, credit constraints are nonbinding.

household is \overline{Y} in each period leading to utility $u(\overline{Y})$ in each period. The household buys if and only if (we drop the subscripts *i* from this point on to avoid clutter):

(1)
$$u(\bar{Y}) - u(\bar{Y} - \frac{P}{2}) \le R.$$

Decreasing marginal utility implies that $u(\bar{Y} - \frac{P}{2})$ increases faster than $u(\bar{Y})$ in income, meaning that acquisition is increasing in income.

With credit constraints, we consider the three possible cases: the consumer buys in period 1, buys in period 2, or never buys. We'll define the trade-offs involved in each of these choices and plot the implied indifference surfaces to show how the propensity to purchase changes with income.

Case 1 (Buy in Period 1).—The household is indifferent between buying in period 1 and not buying if and only if:

(2)
$$\frac{u(Y_1) - u(Y_1 - P)}{2} = R$$

Define the surface *B* as all (Y_1, Y_2, R) triplets that satisfy this equality. Households above this surface strictly prefer to buy in period 1 than to not acquire.

Case 2 (Wait and Buy in Period 2).—The household is indifferent between saving and buying in period 2 and not buying if and only if:

(3)
$$u(Y_1) + u(Y_2) - 2u(\bar{Y} - \frac{P}{2}) = R.$$

Define the surface *W* as all (Y_1, Y_2, R) triplets that satisfy this equality. Households with *R* greater than this strictly prefer to buy in period 2 than to not acquire.

This comparison provides intuition as to how credit constraints affect the acquisition decision. Decreasing marginal utility implies that consumers gain from equating consumption across periods. An unconstrained household can buy in period 1 and equate consumption in each period through borrowing. Credit constrained households cannot. They can create or magnify inequality in consumption between periods and buy immediately, or give up some of the benefit of the asset by delaying the purchase, but gain by being able to equate consumption across periods by saving. Higher income in any period leads savers to purchase more, but first period buyers respond only to their first period income, because they cannot access second period income to purchase in the first period (Lemma 1 below). Thus, an increase in cumulative income increases acquisition through delay or saving (Lemma 2 below).

LEMMA 1: From decreasing marginal utility:

- (*i*) *B* is decreasing in Y_1 and constant in Y_2 .
- (*ii*) W is decreasing in Y_1 and Y_2 .

PROOF:

Note that

- (4) $\frac{\partial B}{\partial y_1} = \frac{u'(Y_1) u'(Y_1 P)}{2}$
- (5) $\frac{\partial B}{\partial y_2} = 0$
- (6) $\frac{\partial W}{\partial Y_1} = u'(Y_1) u'\left(\bar{Y} \frac{P}{2}\right)$
- (7) $\frac{\partial W}{\partial Y_2} = u'(Y_2) u'\left(\bar{Y} \frac{P}{2}\right).$

Signs follow directly from decreasing marginal utility.

LEMMA 2: Now we consider the joint distribution of R and \overline{Y} . Fix a. If there exist incomes and valuations such that some households save to buy and some households buy immediately, then there is at most a single intersection of B and W such that for low average incomes (below the intersection) the marginal acquiring household buys in period 1 (and has correspondingly high valuations), while for higher incomes the marginal acquiring household saves and buys in period 2.

PROOF:

Since both *B* and *W* lines are monotonically nonincreasing in Y_1 and Y_2 it is sufficient to show that *W* is steeper. Note that $Y_1 = (2 - a)\overline{Y}$ and $Y_2 = a\overline{Y}$, and $a \in [1,2]$, since income is weakly increasing between periods. From Lemma 1, we have that $\frac{\partial W}{\partial Y_2} < \frac{\partial B}{\partial Y_2} = 0$. So, provided that $\frac{\partial W}{\partial Y_1} - \frac{\partial B}{\partial Y_1}$ is bounded above, there is a high enough *a* for this condition to hold. $\frac{\partial W}{\partial Y_1} - \frac{\partial B}{\partial Y_1}$ is bounded above under a variety of standard assumptions on the third derivative. Online Appendix A outlines these.

Figure 4 depicts the relationship outlined in Lemma 2 in \overline{Y} , R space. We hold the ratio of Y_2 to Y_1 fixed, which means plotting the indifference surfaces as lines for a given value of a. The dotted line B is the set of households who are indifferent between buying immediately and never buying. The solid line W, is the set of households who are indifferent between saving in period 1, buying in period 2, and never buying. Households above either line acquire assets.¹⁵

COROLLARY TO LEMMA 2: There is an S-shaped curve in ownership at the end of period 2.

Simply flip the y-axis in Figure 4 and note that those with valuations below the lines do not acquire. The intersection defines a kink. For incomes below the kink,

¹⁵Note that households above both lines either save or buy. For each income level, there is a cutoff in R such that households above it buy immediately, while households below it, if any, save.



FIGURE 4. EXAMPLES OF NONLINEAR RELATIONSHIP BETWEEN INCOME AND ENERGY USE

Notes: Figure 4 represents the relationship outlined in Lemma 2. Households above indifference curve B strictly prefer to buy in period 1 than to not acquire. Households above indifference curve W strictly prefer to buy in period 2 than to not acquire.

acquisition thresholds rise slowly (the line *B* is relatively flat). At the kink, the rate of acquisition increases as households move to the steeper *W* line.

Lemma 3 below follows from another bit of intuition. If second period income is higher than first period income, income growth has effectively forced the household to delay some consumption. Call this "forced savings." For a household who is not saving or purchasing in the second period this reduces welfare because the marginal value of consuming that income in the second period is lower than in the first. However, if the household saves to purchase, the household is not subject to this unequal consumption of the non-asset goods, so the welfare difference increases between saving and purchasing and not purchasing or purchasing immediately. This increase can induce "complimentary savings" where once the household is forced to partially save, the household now might choose to have additional savings.

LEMMA 3: Hold cumulative income fixed (\overline{Y}) , then increasing period 2 income (Y_2) reduces period 1 purchasers and increases savers/period 2 purchasers.

PROOF:

(8)
$$\frac{\partial B}{\partial a} = - \bar{Y} \frac{\partial B}{\partial Y_1},$$

which is positive from decreasing marginal utility, so there are fewer who buy immediately.

(9)
$$\frac{\partial W}{\partial a} = \bar{Y} \left(u'(a\bar{Y}) - u'((2-a)\bar{Y}) \right),$$

which is negative if income is growing a > 1, so there are more savers.

Lemma 3 is effectively a point-wise comparison in Figure 4; shifting income from the first period to the second shifts the line B up and the line W down. Lemma 4, below, shows that the magnitude of the shifts in B and W are greater at higher incomes.

COROLLARY TO LEMMA 3: Income growing fast enough is a sufficient condition for the increase in savers to outweigh the decrease in buyers, and thus, for there to be more total purchasers.

PROOF:

For there to be more total purchasers, the corollary requires that the increase in savers outweigh the loss of immediate buyers. Note that if $a \rightarrow 2$ there are no immediate buyers since first period income does not support immediate purchase. As such, if income is growing fast enough, the corollary holds.

LEMMA 4: The effects of Lemma 3 are increasing in income: at higher cumulative income (\bar{Y}) , the decline in the range of R that corresponds to period 1 purchasers is greater and the range of R that corresponds to savers/period 2 purchasers increases at higher cumulative income levels, as long as income growth is fast enough.

PROOF:

See online Appendix A.

While the model presented in the text is stylized to focus on the clear intuition, we present additional results and discuss a number of important extensions in online Appendix A. We show that our results are robust to complementarities between durables and utility from nondurables and to uncertainty about future income. Indeed, both mechanisms magnify certain results of the model. We also demonstrate that our results, including the nonlinearities we predict, are robust to price changes, multiple assets, and to adding more periods.

C. Model Predictions

There are a number of empirically testable predictions from this model.

Prediction 0.—*Acquisition is increasing in income.* With credit constraints, this follows from Lemma 1. Without credit constraints, this follows from the declining marginal utility of income.

Prediction 1.—There is an S-shaped curve in acquisition. Without credit constraints, this follows for reasonable distributions of R. This unconstrained setup is consistent with Farrell (1954) and Bonus (1973), who assumed distributions of valuation parameters and income thresholds, respectively, and showed that these lead to S-shaped logit or probit curves for appliance ownership. With credit constraints, this is the corollary to Lemma 2. Note that the mechanisms are orthogonal. The credit constraints mechanism does not depend on distributions of valuations or exogenous income thresholds. This translates into an S-curve in current income, as plotted in Figure 1, for example, as long as there is some positive correlation in period-to-period income. Prediction 2.—If there are relatively few period 1 buyers, faster income growth leads to delays in acquisition and more ownership at the end of period 2. Recall that if there are no credit constraints, then for a fixed \overline{Y} faster income growth has no impact on the number of purchasers. There are relatively few period 1 buyers, for example, if incomes are low or refrigerators have low value or are expensive relative to incomes. This prediction, like Lemma 3, is effectively a point-wise comparison in Figure 4 above. It shifts B up and shifts W down. Note that showing an S-shape is not a precise test of the existence of credit constraints because we can get it with either distributions of preferences or credit constraints. This provides another way to validate the importance of credit constraints empirically.

Prediction 3.—*Ownership after period* 1 *is increasing in the interaction of income level and growth.* Follows from Lemma 4.

III. Empirical Context

To better understand the growth in energy demand among low-income households, we analyze asset acquisition in the context of Oportunidades, a conditional cash transfer program in Mexico that was designed to break the intergenerational transmission of poverty. The Oportunidades program is well studied. See Gertler, Martinez, and Rubio-Codina (2012) and Parker and Todd (forthcoming) for additional description. Here we focus on the specifics most relevant to our empirical analysis.

A. Program Benefits

Cash transfers from Oportunidades are given to the female head of the household every two months conditional on two criteria. First, all beneficiary households receive a fixed stipend as long as family members obtain preventive medical care and attend pláticas, or educational talks on health-related topics. Second, households also receive educational scholarships conditional on children attending school a minimum of 85 percent of the time and not repeating a grade more than twice. The educational stipend is provided for each child younger than 18 years old enrolled in school between the third grade of primary school and the third grade of high school (twelfth grade) and varies by grade and gender. It rises substantially after graduation from primary school and is higher for girls than boys during high school. Only children who were living in the household when the program started are eligible for the school transfers in order to prevent migration into the household. Total transfers for any given household are capped at a predetermined upper limit.¹⁶

Table 2 describes the benefits to which beneficiary households were entitled in 2003. While the benefit levels and the grades covered have changed over the course of the program, its basic structure has not. In 2003, the basic support was 155 pesos every two months. The educational scholarship in 2003 ranged between 105 pesos

¹⁶Compliance was verified through the clinics and schools, who certified whether households actually completed the required health care visits and whether kids attended schools. While full compliance varied, only about 1 percent of households were denied the cash transfer completely for noncompliance.

Basic Support	155	
Educational scholarship		
Grade	Boys	Girls
Third	105	105
Fourth	120	120
Fifth	155	155
Sixth	205	205
Seventh	300	315
Eighth	315	350
Ninth	335	385
Tenth	505	580
Eleventh	545	620
Twelfth	575	655

TABLE 2—OPORTUNIDADES BIMONTHLY SUPPORT LEVELS IN 2003 (Pesos)

Notes: A household can receive a maximum of 1,025 pesos with children through sixth grade or 1,715 pesos with children in seventh grade or higher. An additional 200 pesos for children in third to sixth grades and 250 pesos for children in seventh grade or higher are provided once per year for school supplies.

for children in the third grade to 655 pesos for teenage girls in twelfth grade. Finally, Oportunidades provides a yearly stipend to cover the costs of school supplies for children who do not get them at school.

As Table 2 documents, differently composed households are eligible to receive different transfer amounts. For example, households with more female children enrolled in higher grades are eligible for larger educational stipends than similar households with children enrolled in lower levels or with more male children. We can compute the maximum potential transfer for a family by applying the values from Table 2 to the following formula:

(10)
$$PT_{it} = \min(T_t^{max}, BT_t + \sum_s ST_{st}NK_{sit}),$$

where PT_{it} is the maximum potential transfer that could be received by household *i* in period *t*, T_t^{max} is the program cap on benefits, BT_t is the basic transfer amount that all households receive (the food support), ST_{st} is the educational transfer conditional on a child of type *s* (i.e., based on grade and gender) attending school, and NK_{sit} is the number of children of type *s* in household *i* at baseline aged forwarded to period *t*. Because of both the benefit schedule and the cap on total benefits, potential transfers are a nonlinear function of the number of children at baseline who could attend the grades eligible for the educational scholarships in period *t*.

The actual transfers received by a household are less than the potential amount if some children do not attend school. Thus, the actual bimonthly transfer amount received by household i at each time *t* is computed by using a version of equation (10) that replaces NK_{sit} with a variable measuring the number of children of type *s* in household *i actually* attending school in period *t*.

B. Oportunidades Evaluation and Data Collection

The data used in this study were generated for program evaluation. At the outset of the program, the government identified marginalized communities in seven states in central Mexico and randomly allocated 320 to be early intervention and

1381

186 late intervention. The original classification scheme designated approximately 52 percent of households as eligible ("poor") (Hoddinott and Skoufias 2004). In the marginalized communities, credit is scarce. Only 1.2 percent of the villages have formal credit institutions, and only 2.8 percent of eligible households have loans of any kind, mostly from friends, neighbors, and relatives (Gertler, Martinez, and Rubio-Codina 2012).¹⁷

Eligible households in the early intervention communities received benefits starting in April 1998, while households in the late intervention communities did not receive benefits until October 1999. Eligible households were offered Oportunidades and a majority (90 percent) enrolled in the program (Gertler, Martinez, and Rubio-Codina 2012). Once enrolled, households received benefits for at least a nine-year period conditional on meeting the program requirements. No sites were told in advance that they would be participating in the program, information about timing of program rollout was not made publicly available, and there is no evidence of anticipatory behavior (Attanasio, Meghir, and Santiago 2012).

Our analysis focuses on these 506 communities and the panel of approximately 10,000 households that were surveyed from 1997 through 2007. Treatment and control households, which we will refer to as "early" and "late," were similar on a wide array of measured characteristics. Online Appendix Tables 1 and 2 summarize a number of different household-level attributes separately for early and late households. For nearly all of the variables, the means are statistically indistinguishable across the two groups, suggesting that the randomization successfully created comparable groups.

The data used in this study come from the baseline survey, the Encuesta de Características Socioeconómicas de los Hogares (ENCASEH), and the Oportunidades Evaluation Survey (ENCEL), which is a panel dataset. We use all survey rounds in which data were collected about household assets: the first survey, which was administered a year after the program started, during the fall of 1998 and additional surveys from fall 1999, spring and fall 2000, fall 2003, and fall 2007.

The evaluation surveys gather information on a number of potential metrics that the program may affect, including household and household members' characteristics, income and labor supply, expenditure, health and nutritional status, and education, among others. Of particular importance for this study, the survey gathers information on energy-using household durable asset possession, such as refrigerators, gas stoves, televisions, and washing machines.

IV. Empirical Specification and Identifying Assumptions

In this section, we specify the empirical models that we use to test predictions from our model of the relationship between income and durable asset accumulation. Since durable asset purchases are infrequent, discrete events, we model the decision as the probability of purchase in a particular period given that the household has not yet purchased the asset. As such, we estimate linear discrete-time hazard models that take advantage of the panel structure of our data.

¹⁷Because future income streams from Oportunidades were quite certain relative to income from general economic development, we suspect that credit constraints might bind even more in other environments.

We first test Prediction 0, that the probability of asset purchase is an increasing function of cumulative income, in the context of the following specification:

(11)
$$h(a_{it}) = \Pr(a_{it} = 1 | a_{it-1} = 0)$$
$$= \gamma_0 + \alpha_1 Cumulative Income_{it} + \beta \mathbf{X}_i + \beta_t \mathbf{F}_i + \mathbf{R}_{rt} + v_{it},$$

where $h(a_{it})$ is the probability that household *i* acquires appliance *a* in period *t*, conditional on not having it in period t - 1; \mathbf{X}_i is a vector of control variables including household characteristics measured at baseline, which we sometimes replace with a household fixed effect (λ_i) ; \mathbf{F}_i is a vector of family structure variables, which we interact with round to allow the relationship to vary over time, for example as children age; and \mathbf{R}_{rt} is a vector of state-by-period dummies that help account for any state-specific changes in asset¹⁸ and electricity¹⁹ prices.²⁰ v_{it} is the random error term. Under Prediction 0, α_1 is positive.

We next test Prediction 1 that there is an S-shaped curve in acquisition. In our context, where all the households are poor, an S-shape implies that higher income households are more likely to use the same increases in income to purchase an appliance. We test this by estimating:²¹

(12)
$$h(a_{it}) = \Pr(a_{it} = 1 | a_{it-1} = 0)$$

= $\gamma_0 + \alpha_1 Cumulative Income_{it} + \alpha_2 Cumulative Income_{it}^2 + \beta \mathbf{X}_i + \mathbf{R}_{rt} + v_{it}$

where Prediction 1 suggests that α_2 is positive.

Finally, we test Predictions 2 and 3 that, conditional on having the same level of cumulative income, a household which accumulated the income more slowly is less likely to acquire an asset and that this effect is increasing in cumulative income. We use the variation generated from the random assignment of households to begin receiving transfers earlier versus later to test these predictions. We include in the model the variable *Early_i*, a dummy indicating that the household began receiving Oportunidades transfers 18 months before the control households, and an interaction of *Early_i* with *Cumulative Income_{ii}*²²

¹⁸ In general, refrigerators are produced on a global market and often imported into Mexico, so prices may largely reflect global trends. Moreover, Oportunidades households are a small share of durable purchases and buy from large retailers in distant urban centers, so prices likely did not respond to transfers.

¹⁹Mexican electricity tariffs are set at the state level.

²⁰Note that previous related papers have examined the impact of income on appliance acquisitions (Dubin and McFadden 1984) and ownership (Dargay, Dermot, and Sommer 2007) using cross-sectional variation in income with limited controls for household demographics. In those specifications, unobserved differences across households may be correlated with income and taste for appliances. One substantial advantage of our empirical setting is that we can take advantage of the large shocks to income that households received via the transfers, and we use both within-household differences brought on by the nonlinear transfer schedule and cross-household difference driven, among other things, by randomization.

²¹We use the same notation for the constant and the coefficients on the control variables in equations (11) to (13), but do not in fact constrain these to be the same across equations.
²²It is important to note that we are not evaluating the impact of Oportunidades by comparing households in

²² It is important to note that we are not evaluating the impact of Oportunidades by comparing households in treated and control villages. Instead, we are interested in how the level and timing of transfers affect asset acquisition. The randomization provides exogenous variation in the timing of transfers, but because we also model the effect of cumulative transfers directly, we are not simply comparing treated with control households. To avoid confusion, we have relabeled treated households "early" and control households "late."

(13)
$$h(a_{it}) = \Pr(a_{it} = 1 | a_{it-1} = 0) = \gamma_0 + \alpha_1 Cumulative Income_{it}$$

 $+ \alpha_3 Early_i + \alpha_4 Early_i \times Cumulative Income_{it} + \beta \mathbf{X}_i + \mathbf{R}_{rt} + v_{it}$

This specification allows us to test Predictions 2 and 3, which suggest that both α_3 and α_4 will be negative.

A. Measurement of Cumulative Income

We measure *Cumulative Income*_{*it*} as the sum of earned wage income for household *i* in time *t*, own enterprise (farm and nonfarm) income, Oportunidades transfers, and non-Oportunidades public and private transfers. The key issue in analyzing this income variable is measurement error that might lead to attenuated and imprecise estimates. Since surveys typically gather information about current income flow, rather than cumulative income, we have to interpolate income between survey rounds adding to the noise in our estimates.²³ In addition, as is common in the developing world, households have substantial informal and enterprise earnings that can suffer from reporting biases.

In addition to measurement error there is a double counting issue if household invest some of the Oportunidades transfers in enterprises that produce income in future periods and we do not observe the costs of those investments. The presence of investment effects may bias the estimated income effects downward.²⁴ Gertler, Martinez, and Rubio-Codina (2012) do indeed find that households invest a portion of the transfers in agricultural production, providing increases in future income. In online Appendix C, we present results that suggest this creates little, if any, bias.

B. Cumulative Oportunidades Transfers

An alternative is to use the variation in cumulative Oportunidades transfers to generate cross-household variation in income. Administrative Oportunidades data provide clean, comprehensive information on the amount and timing of individual transfers. Transfers result in a dollar-for-dollar increase in income if there is no effect of transfers on nontransfer income, an assumption that we discuss and test at length below.

In all but equation (12) we can replace *Cumulative Income*_{it} with *Cumulative Transfers*_{it}. However, in (12), it is difficult to detect a convexity in response to cumulative income through a squared term of only one component of income. Instead, we estimate a specification that interacts *Cumulative Transfers*_{it}.

²³ In many empirical settings, including some of the authors' own work, consumption is used as a measure of household income because it is reported with more precision. However, as we are estimating a model of savings, using consumption as a proxy for income is not appropriate. Indeed the complementary savings effect would cause a relationship between lower consumption and increased asset acquisition.

²⁴ For example, if a household invests 100 pesos of transfers in 1999 into an endeavor that yields 120 pesos in 2000 (such as farm assets) we will observe the household earning both the 100 pesos in 1999 and the 120 pesos in 2000. Because our model and estimation are in terms of cumulative income, we would measure this household's cumulative income as 220 pesos higher than a household which did not receive the transfer. This will lead us to underestimate the coefficients on income because this household has the same underlying propensity to buy a refrigerator but 100 pesos less in cumulative income than we measure.

JUNE 2016

with dummy variables indicating whether a household started with low or high assets:

(14)
$$h(a_{it}) = \Pr(a_{it} = 1 | a_{it-1} = 0)$$

= $\gamma_1 + \alpha_5 Low Assets_i \times Cumulative Transfers_{it}$
+ $\alpha_6 High Assets_i \times Cumulative Transfers_{it} + \beta \mathbf{X}_i + \mathbf{R}_{rt} + v_{it}$

where *Low Assets*_i is an indicator that the household in is in the lower three-quarters of the asset distribution at baseline and *High Assets*_i is an indicator that the household in is in the top quarter of the asset distribution. Under Prediction 1, $\alpha_5 < \alpha_6$.

Replacing Cumulative Incomeit with Cumulative Transfersit will lead to unbiased estimates of the income effects if the net impact of transfers on total income is one, i.e., nontransfer income is uncorrelated with transfers. Existing work has examined a number of components of household income that may be correlated with transfers. The literature has found labor supply effects of transfers on youth age 12-14 with relatively low wages,²⁵ consistent with some of the transfers being conditional on these age children attending school. However, there is no evidence that adult labor supply responds to transfers (Parker and Skoufias 2000 and Skoufias and Parker 2001). Although Albarran and Attanasio (2004) find that transfers lead to statistically significant but small reductions in private remittances to households that receive remittances, only 7 percent of households receive remittances at all. Finally, Gertler, Martinez, and Rubio-Codina (2012) report that households invest a portion of the transfers in agricultural production, but that the size of the investment effect on income is also small (see online Appendix C).²⁶ Overall, they show that the marginal propensity to consume transfers is 74 percent, which is consistent with the marginal propensity to consume income in other developing countries (Gertler, Martinez, and Rubio-Codina 2012) and thus consistent with a net impact of transfers on total income of one.

Also, since the cumulative transfers that a family receives are determined by choices about whether to keep children in school, it is conceivable that the decision to purchase an appliance would be correlated with household-level shocks that altered the parameters of these choices. For instance, if the household experienced a large positive shock to its nontransfer income, it might be more likely to leave children in school instead of working and simultaneously make the household more likely to acquire an appliance. This would lead to a positive bias in the coefficient on *Cumulative Transfers*.

²⁵ Parker and Skoufias (2000) and Skoufias and Parker (2001) document that the program reduces child labor and increases enrollment in junior high (secondary) schools as the opportunity cost of these children being in the labor force is now higher. Schultz (2004) also finds positive effects for primary school and junior high school enrollment for boys and girls.

²⁶ If households invest transfers, then our estimate of the income effect from cumulative transfers specifications are too high. Continuing the example from footnote 24, we attribute the household's response to 100 pesos in transfers, rather than the 120 pesos in income the household actually received. The transfer specifications therefore provide a lower bound and the income specifications provide an upper bound, though in practice both bounds are qualitatively similar and are consistent with our predictions.

C. Instrumenting with Potential Transfers

We estimate separate models using *Cumulative Income* and *Cumulative Transfers* and therefore need to address both the measurement error in *Cumulative Income* and potential omitted variable bias in *Cumulative Transfers*. We solve both of these problems through instrumental variables using *Potential Cumulative Oportunidades Transfers* (PCT), the transfers a household had the potential to earn if they enrolled and fully complied, as the instrument. It is computed assuming that all eligible children enrolled at baseline are still in school and have advanced one school grade per school year, and therefore by construction is uncorrelated with adult and child labor supply, enterprise income, and other nonlabor income.

There is exogenous variation in PCT both within a given household over time and across households. The cross-sectional variation at a point in time depends on when the family entered the program and the rate of accumulation since entry. While the time the household was incorporated into the program was randomized, the rates at which potential cumulative transfers change over time are a nonlinear function of school grade and gender of school-age children.

The nonlinearities arise from the program rules described above. Rates of accumulation within a household vary with time as younger children age into the program, as they progress through school, and as children age out of the program. There are very large nonlinearities as children enter fourth grade, secondary school, high school, graduate high school, and based on gender. In addition, in 2000, Oportunidades extended the payments for grades 10–12. These rules generate substantial exogenous nonlinear variation in the transfer amounts across households as well as within households over time. PCT is a sufficient statistic for all of these nonlinearities. Differences in household demographics and initial program participation dates drive differences in PCT between households. As households cross the various nonlinear payment thresholds, PCT picks up differences within households over time. Moreover, the cap on total per-period benefits creates another nonlinearity that interacts with all of the other nonlinearities.²⁷

To better understand the extent to which different factors drives variation in PCT we decomposed the variance as follows:

(15)
$$\operatorname{var}(PCT_{it}) = \operatorname{var}(\delta_i) + \operatorname{var}(\mathbf{R}_{rt}) + 2\operatorname{cov}(\delta_i, \mathbf{R}_{rt}) + \operatorname{var}(\varepsilon_{it}),$$

where δ_i and R_{rt} represent the household and region-by-period fixed effects, respectively, and ε_{it} is a random error term. Our calculations suggest that region-by-period fixed effects account for about 55 percent of the variation, household fixed effects account for about 30 percent of the variation, and the covariance between the two terms is effectively zero. This suggests that there is substantial remaining variation

²⁷ An alternative identification strategy would try to exploit one or two specific nonlinearities using ad hoc specifications. We reject this for several reasons. First, the cumulative transfer formula is the parameterization of the instrument that incorporates all of the nonlinearities and their interactions. Since this parameterization is the one actually faced by families and summarizes all the nonlinearities, then it should have more predictive power than other limited ad hoc parameterizations. Second, by focusing on one or two nonlinearities we would be ignoring other relevant variation and we would lose potentially substantial predictive power.

within household to help identify the effects of transfers in the specifications with household and region-by-period fixed effects.

Because some of our specifications rely on cross-household variation, we also estimated the share of the variance accounted for by household factors that might reasonably impact appliance valuations. When we include indicator variables for household size and the age structure of household members instead of household fixed effects, these variables explain less than 15 percent of the variation in transfers. These decompositions suggest that the randomization, differences in transfers driven by household gender and age-composition of children, and the nonlinearities in these transfer schedules account for a substantial share of our variation.

One potential concern is that potential cumulative transfers may be correlated with household demographic structure and the demographics may affect the demand for energy-using assets. We address this by explicitly controlling for household demographic structure in the models. Also, note that potential cumulative transfers are not strongly correlated with the number of children in the household. To see this, compare a household with three girls in grade two of primary school and a household with three girls in grade eight (junior high school). Both households have three female children but while the first household will receive no school transfers in the current period, the latter household will receive a large monthly transfer. In addition, families with four or more children in junior high school would receive the same transfer amount as the latter household because the cap on total benefits would be binding. Thus, we are able to explicitly control for household size and the number of children in the household in the empirical specification and still have substantial variation in cumulative transfers to identify the coefficient on that variable.

There may be lingering concerns that household demographic structure is correlated with both the potential cumulative transfer amounts and the propensity to purchase appliances. For instance, households with older girls may have systematically different refrigerator valuations than households with younger girls. We explicitly test to see whether baseline appliance ownership is correlated with future cumulative transfers and therefore with household demographic structure embedded in the transfer formula. Specifically, we will present results from placebo tests that suggest that the nonlinear function that translates family structure to potential cumulative transfers does not predict appliance ownership at baseline (i.e., before the program started). So, as long as any changes in the propensity to buy a refrigerator after the baseline survey were similar across households with different family structures, our specification will yield unbiased estimates of our coefficients of interest.

D. Interaction Terms

In equation (13) some specifications include the interaction $Early_i \times Cumulative Transfers_{it}$ in order to measure differences in refrigerator acquisitions between early and late households who have had the same level of cumulative transfers. While variation in $Early_i$ in equation (13) is generated by the random assignment of households to begin receiving transfers early versus late, we also want to be sure that the distributions of cumulative transfer levels overlap between the early and late groups. Otherwise, we could simply be picking up nonlinearities in the relationship between cumulative transfers and appliance acquisition.

	Late households			Early households			
	25 percent	Median	75 percent	25 percent	Median	75 percent	
1998				0.09	0.13	0.21	
1999				0.24	0.38	0.61	
2000m	0.06	0.09	0.16	0.32	0.51	0.82	
2000n	0.18	0.33	0.53	0.45	0.79	1.23	
2003	0.93	1.69	2.63	1.24	2.19	3.36	
2007	2.36	3.89	5.67	2.63	4.33	6.35	

TABLE 3—SUMMARY STATISTICS: CUMULATIVE TRANSFERS (Ten Thousands of 2003 Pesos)

Notes: Data from the survey administered in spring 2000 is denoted by 2000m. Data from the survey administered in fall 2000 is denoted by 2000n.

Source: ENCEL surveys and administrative data

Table 3 reports cumulative transfer amounts over time for households in the early and late groups that are at different parts of the transfer distribution. We see that households at the seventy-fifth percentile of the late group had higher cumulative transfer amounts than both households at the twenty-fifth percentile of the early group by late 2000 and the median of the early group by 2003. This suggests that we have considerable overlap between the distributions of cumulative transfers by 2003.²⁸

In equation (14) we interact *Cumulative Transfers_{it}* with the indicator *Low Assets_i* in order to test heterogeneous treatment response of wealthier households as predicted by the model. Variation in baseline *Low Assets_i* reflects households' pretreatment wealth and income, and may have various endogeneity concerns. For example, households with more animals may simply value assets more. However, in all our relevant specifications, we control for a household's baseline asset ownership.²⁹

V. Empirical Results

Our analysis focuses on refrigerator acquisitions, by far the most expensive and most energy-intensive household appliance for the Oportunidades population. The results for refrigerators are robust for other assets as we discuss below. In addition, while we report specifications estimated using both *Cumulative Income* and *Cumulative Transfers*, we favor the results estimated with transfers. The results are very similar for both, but transfers are measured with much more precision.

A. Descriptive Evidence

We begin with descriptive support for the theoretical predictions. Figure 5 plots cross-sectional refrigerator ownership as of 2003 on *Cumulative Transfers* through

²⁸We focus on observations through 2003. As the initial randomization is the only source of difference between early and late households, additional years mute the variation between the two groups. If we include later years in our estimates of equation (13), the coefficient estimates are attenuated to zero slightly, as we would expect with more noise relative to systematic differences between the groups, but are very similar to the results we present.

²⁹ In the models reported in the paper we use the value of animal assets to measure *Low Assets*_i. Alternate measures of household wealth, such as total household consumption or consumption per capita, yield similar differences between the relatively better off and less well-off households.



FIGURE 5. REFRIGERATOR OWNERSHIP AND CUMULATIVE TRANSFERS IN 2003

Notes: LOWESS regressions. Excludes the bottom and top 2 percent of cumulative transfers in each group. *Sources:* ENCASEH (1997) and ENCEL (1998, 1999, March 2000, November 2000, 2003, and 2007) surveys and administrative data. See online Appendix. Reflects only households who do not have refrigerators at baseline.

2003 separately for *Early* and *Late* households.³⁰ Following Prediction 0 we expect upward-sloping ownership curves ($\alpha_1 > 0$). Prediction 1 predicts that these curves are convex, i.e., increasing in other components of wealth ($\alpha_2 > 0$). Prediction 2 suggests that the line for *Early* households is below the line for *Late* households ($\alpha_3 < 0$), while Prediction 3 implies that the line for *Early* households is less steep than the line for *Late* households ($\alpha_4 < 0$). Figure 5 is consistent with these predictions.³¹

B. Income Effects

Table 4 presents the results for several specifications of equation (11) that test Prediction 0. We estimate (11) using linear regression and report robust standard errors clustered at the village level. All specifications include state-by-round fixed effects and either a number of household controls or household fixed effects (FE). The first three columns report results based on *Cumulative Transfers_i* while the last three report results based on *Cumulative Income_i*.

The coefficient on *Cumulative Transfers* (α_1) in the first three columns is positive and highly statistically significant. Column 1 reports the OLS estimate, column 2 reports the instrumental variable (IV) estimate with household controls using *PCT* to instrument for *Cumulative Transfers*, and column 3 reports the IV estimates replacing the household controls with household fixed effects. The instrument is

³⁰The survey only collected information on whether the household has a refrigerator at the time of the survey and not on the type of refrigerator owned.

³¹Though there are convex regions, the convexity of Prediction 1 is not obvious in the figure. In Section VC we test for convexity explicitly including controls.

	Discrete time hazard		Household FE	Discrete time hazard		Household FE
	OLS (1)	IV (2)	IV (3)	OLS (4)	IV (5)	IV (6)
Cumulative Transfers	0.018*** [0.005]	0.020*** [0.007]	0.047*** [0.008]			
Cumulative Income				0.003*** [0.001]	0.016*** [0.005]	0.034*** [0.007]
Observations R^2	30,414 0.103	30,414	30,258	30,414 0.104	30,414	30,258
Kleibergen-Paap Wald F-Stat		2,503	2,060		92	108
Number of households			6,655			6,655

TABLE 4—BASIC RESULTS: REFRIGERATOR (Income Effects)

Notes: All specifications include state by round-fixed effects. Rounds through 2003 included. All specifications include household age structure controls including number of children 7 and younger, number of children 8 to 17 years old, number of males 18 to 54, number of females 18 to 54, number of adults 55 and older, number of individuals with unreported ages, interacted with round fixed effects. Specifications in columns 1, 2, 4, and 5 include head of household's gender, head of household's age and education, and whether the household owns the house they live in, farm assets at baseline, number of other social programs the household is the beneficiary of, and village characteristics including migration intensity, marginalization, and distance to nearest city. In column 2, 3, 5, and 6 the excluded instrument is PCT. Columns 3 and 6 include household fixed effects and drop 156 singletons. Robust standard errors clustered by village in brackets.

***Significant at the 1 percent level.

** Significant at the 5 percent level.

*Significant at the 10 percent level.

extremely strong as indicated by the first-stage *F*-statistics, which exceed 2,000. The coefficient estimates in (2) are very similar to the OLS estimates in (1). However, the IV FE estimate in (3) is more than twice as big as the OLS and IV estimates in (1) and (2).³²

The magnitude of the estimated coefficient in column 3 suggests that for every 10,000 pesos increase in a household's cumulative transfers, the probability that it acquires a refrigerator goes up by 4.7 percentage points off of a baseline ownership of 4 percent.³³ Mean cumulative transfers in 2003 explain 55 percent of refrigerator acquisition between baseline and 2003.

Columns 4 through 6 repeat these estimates using *Cumulative Income*. The large magnitude difference between column 4 and 5 is consistent with the IV approach correcting for the much larger measurement error in income. The coefficients estimated by IV using *Cumulative Transfers* in columns 2 and 3 and those estimated using *Cumulative Income* in columns 5 and 6 are similar in magnitudes and significance.³⁴ One important difference, however, is in the power of the first stage. The

³² It is possible that the coefficient on *Cumulative Transfers* is larger in column 3 in part because the household fixed effects capture whether the household was in the early or late wave of program treatment, which we show below is an important factor in determining refrigerator purchases.

³³We opt not to estimate a traditional income elasticity. Indeed, as our conceptual framework suggests, the likelihood of acquiring an asset with an increase in current income depends not only on that income, but on past income accumulated as wealth and the time path of future income. Unlike in stable settings with well-functioning capital markets, current income is not a good proxy for a household's expected path of income.

³⁴We discussed above why measurement and data issues make the transfers effect potentially too large, and the income effect too small.

F-statistics are substantially lower in the income specifications in columns 5 and 6 compared to the transfers specifications in columns 2 and 3.

C. Nonlinearity in Income and Wealth Effects

We now test the hypothesis that the income effects are increasing in income (Prediction 1), using the specification in equation (12) that includes both *Cumulative Income* and *Cumulative Income*^{2,35} As instruments we use *PCT* and *PCT* interacted with the value of assets at baseline.³⁶ The results, reported in Table 5, indicate a convex relationship between cumulative income and asset acquisition; i.e., $\alpha_2 > 0$ in equation (12). The magnitudes suggest that the income effect of the 75 percent percentile household in 2003 is nearly twice that of the median household.³⁷

These results are consistent with an S-shaped relationship between income and asset ownership. Since all households in the sample are poor, at baseline we would expect most of them to be at cumulative income levels below the acquisition cutoff, i.e., the first inflection point on the S-curve. We would also not expect any households to pass the second inflection point. As households enter the program, the slightly better off should be more likely to acquire the asset at a given level of transfers than those that are poorer leading to the first inflection point.

We find similar results when we estimate the specification in equation (14) and allow the effects of transfers on asset acquisition to differ for households in the top 25 percent of asset ownership at baseline versus the bottom 75 percent. This allows us to identify the Oportunidades households that are relatively better off and closer to the acquisition inflection point on the S-curve. Indeed, the results reported in Table 6 show that the effect of transfers on acquisition is larger for households that started off wealthier. The coefficients from column 2 suggest that 10,000 pesos in cumulative transfers leads to a 1.5 percentage point increase in ownership by poor households, while wealthier households increase ownership by 3.1 percentage points. These results are consistent with the importance of credit constraints steepening the S-curve. Income specifications of this interaction, reported in online Appendix C, show similar, though less precise results.

³⁶ Another potential instrument is PCT^2 . In specifications that include this, the power of the first stage is lower and estimates are not statistically different from those reported in Table 5. This is because there is little nonlinearity in the relationship between *Cumulative Income* and *PCT* conditional on the rich set of controls. See online Appendix Figure 1.

 37 Marginal effect at median income (12.4 ten thousand pesos) = $0.0059 + 12.4 \times 0.0009 = 0.0053$. At seventy-five percentile (16.9 ten thousand pesos) = $0.0059 + 16.9 \times 0.0009 = 0.0093$.

³⁵ For this specification we add to each household's cumulative income the value of their animal assets at baseline. We use baseline assets to measure initial savings from cumulative income prior to the period of analysis. In rural Mexico, few households in this population have access to banks. Rural households in low-income countries typically do not hold savings in cash, but rather hold savings in more easily liquefiable assets such as animals (Rosenzweig and Wolpin 1993). As such, we measure the value of assets in our study as the value of animals. Household asset ownership at baseline, then, is a measure of the households' unconsumed previous income. Thus, conditioning on baseline assets, any income that household earned prior to baseline that was not saved, could not influence future consumption and asset purchase decisions. If the primary savings vehicles are these assets, then we have correctly captured the household's wealth at baseline. Baseline animal assets are controls in all specifications, so this change is only relevant for the nonlinearity.

	Discrete tir	Discrete time hazard		
	OLS	IV	IV	
	(1)	(2)	(3)	
Cumulative Income	-0.0007 [0.0011]	-0.0059 [0.0108]	0.0132 [0.0132]	
Cumulative Income ²	0.0001*** [<0.0001]	0.0009** [0.0004]	0.0008** [0.0004]	
Observations R^2	30,414 0.105	30,414	30,258	
Kleibergen-Paap Wald F-Stat on excluded variables		22	23	
Number of households			6,655	

TABLE 5—BASIC RESULTS: REFRIGERATOR (Nonlinear Income Effects)	s)
--	----

Notes: All specifications include state by round-fixed effects and household controls and column 3 includes household-fixed effects as described in the notes to Table 4. In column 2 and 3 excluded instruments are *PCT* and *PCT* × *Baseline Animal Assets*. Robust standard errors clustered by village in brackets.

***Significant at the 1 percent level.

**Significant at the 5 percent level.

*Significant at the 10 percent level.

TABLE (5—BASIC	RESULTS:	Refrigerator	(Wealth	ı Effects)
---------	---------	----------	--------------	---------	-----------	---

	Discrete tin	Discrete time hazard		
	OLS (1)	IV (2)	IV (3)	
Low Assets × Cumulative Transfers	0.014***	0.014***	0.041***	
High Assets \times Cumulative Transfers	0.029*** [0.006]	0.032***	0.058*** [0.009]	
Difference	0.015*** [0.005]	0.018*** [0.006]	0.017*** [0.006]	
Observations R^2	30,414 0.104	30,414	30,258	
Kleibergen-Paap Wald F-Stat		1,252	1,025	
Number of households			6,655	

Notes: All specifications include state by round-fixed effects and household controls and column 3 includes household-fixed effects as described in the notes to Table 4. Columns 1 and 2 also include *High Assets* as a household control. In columns 2 and 3, excluded instruments are *Low Assets* × *PCT* and *High Assets* × *PCT*. Robust standard errors clustered by village in brackets.

***Significant at the 1 percent level.

**Significant at the 5 percent level.

*Significant at the 10 percent level.

D. Timing Effects

Table 7 presents several specifications of equation (13), which we use to test Predictions 2 and 3 related to timing. The first column, repeated from column 1 of Table 4, reports the basic cumulative transfer effect. When we include the *Early* dummy in column 2, the coefficient on *Early* is negative as predicted and highly significant. The magnitude suggests that being in the late group is equivalent to about a 6,000 pesos increase in cumulative transfers for the early group.

		Household FE			
	OLS (1)	OLS (2)	OLS (3)	IV (4)	IV (5)
Cumulative Transfers	0.018*** [0.005]	0.023*** [0.005]	0.035*** [0.007]	0.053*** [0.010]	0.063*** [0.010]
Early		$\begin{array}{c} -0.014^{***} \\ [0.005] \end{array}$	-0.006 [0.005]	-0.008 [0.005]	
Early \times Cumulative Transfers			-0.014** [0.006]	-0.021*** [0.007]	-0.018^{**} [0.007]
Compensating transfer at 2003 median <i>Cumulative Transfers</i> (ten thousand pesos)			1.857** [0.859]	1.727*** [0.599]	
Observations R^2	30,414 0.103	30,414 0.104	30,414 0.105	30,414	30,258
Kleibergen-Paap Wald F-Stat				1,058	1,072
Number of households					6,655

TABLE 7—BASIC RESULTS: REFRIGERATOR (Timing Effect)

Notes: All specifications include state by round-fixed effects and household controls and column 5 includes household-fixed effects as described in the notes to Table 4. Column 1 is the basic cumulative transfer effect repeated from Table 4. Robust standard errors clustered by village in brackets.

***Significant at the 1 percent level.

**Significant at the 5 percent level.

*Significant at the 10 percent level.

When we include the interaction between *Early* and *Cumulative Transfers* in column 3, the interaction term is negative and statistically different from zero, while the coefficient on the *Early* dummy drops in absolute value. As the coefficient on the *Early* dummy in column 3 reflects the early effect at zero cumulative transfers, which is outside the range of our data, we also report the "Compensating Transfer at 2003 Median Cumulative Transfers,"³⁸ which is the amount of additional transfers the median early household would need to have the same acquisition rate as a late household with median cumulative transfers. Graphically, this number maps to Figure 5. Take the median household on the *Early* households line (solid line) and compare it to the *Late* households line (dashed line) at the same Cumulative Transfers. The compensating transfer reflects the distance we would need to move the early household to the right along the solid line until it had the same acquisition rate as the late household.

Column 4 of Table 7 instruments for both *Cumulative Transfers* and *Early* × *Cumulative Transfers* with *PCT* and *Early* × *PCT*. The instruments are extremely strong, and the first-stage *F*-statistic, reported at the bottom of column 4, exceeds 1,000. The IV coefficient estimates are very similar to the OLS estimates in column 3. If anything, the IV coefficient on *Cumulative Transfers* is slightly higher. The coefficients from column 4 suggest that 10,000 pesos in cumulative transfers lead to 5.3 percentage points of acquisition by late households, but only 2.4 percentage points of acquisition by early households.

In column 5, we include household fixed effects, which allow us to control for any remaining differences across households not picked up by the household

³⁸The coefficient estimated is $\gamma_{\tau} = -\frac{\alpha_4 \times MedianTransfersEarly + \alpha_3}{\alpha_1 + \alpha_4}$.

controls included in columns 1 through 4^{39} We have within-household variation in cumulative transfers because of the nonlinear increases in transfers depicted in Table 2 and because children age into or out of the program. We cannot estimate the *Early* dummy as this is a time-invariant household characteristic. The specification in column 5 uses IV estimation, and is therefore comparable to the results in column 4. The coefficient estimates on *Cumulative Transfers* and *Early* × *Cumulative Transfers* are remarkably similar across columns 4 and 5 suggesting that our household controls pick up most cross-household differences in tastes.

Table 7A replicates these specifications using *Cumulative Income* as the endogenous variable. Despite the significantly higher measurement error, the results are qualitatively similar, and the IV specifications appear to at least partially address the measurement error. Perhaps the most straightforward comparison is the compensating transfer that the median early household would need to receive to acquire at the same rate as late households. We estimate the compensating transfer using the coefficients from columns 4 of Tables 7 and 7A to be very close: 17,000 pesos in the transfers specification and 15,000 pesos in the income specification.

Other Assets.—Online Appendix D and online Appendix Table 7 present the main results from Tables 4 to 7 for additional energy-using assets. Our statistical power and the segment of the households included in the specifications both vary by assets. For example, most households have radios at baseline, reducing the estimation sample significantly and focusing the estimation on the poorest households. The results for income, nonlinearities in income, and wealth effects are supportive of our model. For all but two assets, the income, nonlinear income, and wealth effects are all positive as predicted.⁴⁰ The magnitudes of these parameters should not be directly compared because these assets provide different levels of utility and thus increases in income should be expected to produce different increases in acquisition depending on the asset. The comparison of the compensating transfer in the timing effect is comparable across assets. For all but one asset, the compensating transfer in online Appendix Table 7 are statistically indistinguishable from the refrigerator effect, and the one which is distinguishable is Gas Stoves where the timing effect is even larger.

E. Alternative Explanations

While the results presented so far are consistent with the model presented in Section II, they may also be consistent with a number of alternative explanations. In this section we investigate the most obvious ones.⁴¹

³⁹ For example, while the household controls include the number of children, we do not include precise variables measuring their exact gender and age makeup. If across households with the same number of children, the households with older girls, for instance, had higher valuations for refrigerators than the households with younger boys, the coefficient on *Cumulative Transfers* might be biased positive. This could in turn lead to a negative bias on the early dummy as, for a given level of cumulative transfers, the early households are more likely to be comprised of young boys.

⁴⁰ Fan ownership was not collected after 2001, so all the households had received transfers and power is significantly reduced. Gas water heaters show little penetration or acquisition, perhaps indicating that even the higher income households are still too poor to acquire them.

⁴¹For brevity we present transfer specifications. Income specifications are similar.

		Discrete time hazard			
	OLS (1)	OLS (2)	OLS (3)	IV (4)	IV (5)
Cumulative Income	0.003*** [0.001]	0.003*** [0.001]	0.004*** [0.001]	0.031*** [0.007]	0.039*** [0.008]
Early		-0.005 [0.005]	< 0.001 [0.005]	0.005 [0.007]	
Early \times Cumulative Income			-0.001 [0.001]	-0.004** [0.002]	-0.003 [0.002]
Compensating income at 2003 median <i>Cumulative Income</i> (ten thousand pesos)			3.799 [3.715]	1.499** [0.585]	
Observations R^2	30,414 0.104	30,414 0.104	30,414 0.104	30,414	30,258
Kleibergen-Paap Wald F-Stat				38	48
Number of households					6,655

TABLE 7A—BASIC RESULTS: REFRIGERATOR (Timing Effect, Cumulative Income)

Notes: All specifications include state by round-fixed effects and household controls and column 5 includes household-fixed effects as described in the notes to Table 4. In columns 4 and 5, excluded instruments are *PCT* and *Early* \times *PCT*. Robust standard errors clustered by village in brackets.

*** Significant at the 1 percent level.

**Significant at the 5 percent level.

*Significant at the 10 percent level.

Future Expectations.—The variable *Early* may be an indicator for lower expected future transfers and thus the negative coefficient could simply represent lower expected income. Specifically, among early and late households with the same cumulative transfer levels at a given point in time, early households might expect lower transfers in the future since their average transfer rate is lower than the late households.⁴² For instance, late households may simply have more girls than early households who are at the same level of cumulative transfers in 2003.

Table 8 presents results from several specifications that include measures of future transfers as additional explanatory variables. Column 1 of Table 8 reproduces column 3 of Table 7, and then columns 2 and 3 add information about the household's actual future transfers through 2007.⁴³ With rational expectations, realized future transfers proxy for expected transfers. The alternative hypothesis put forward above would suggest that the coefficient on *Future Cumulative Transfers* should be positive. In fact, we find that it is either undetectably different from zero or statistically significant and negative. A negative coefficient is consistent with the intertemporal optimization underlying our framework in Section II, as it suggests that households expecting higher transfers in the future are less likely to buy an asset now, presumably because they are waiting to buy it when their income is higher.

⁴²Because transfer rates vary over time within a household, increasing as younger children enter higher grades and decreasing as older children age out of the program, it is also possible that an early household will have the same cumulative transfers as a late household, but will have *higher* expected transfers. For example, the early household could have begun with younger children, accumulating slowly, while the late household began with older children—accumulating quickly at first, but then slowly later when its children age out of the program.

⁴³ Specifically, *Future Cumulative Transfers* are the amount of transfers the household will receive through 2007 less the transfers the household has already received. We interact this with the round and instrument with *Future PCT*.

	OLS (1)	OLS (2)	IV (3)
Cumulative Transfers	0.035*** [0.007]	0.031*** [0.008]	0.103*** [0.011]
Early	-0.006 [0.005]	-0.005 [0.005]	-0.020^{***} [0.007]
Early $ imes$ Cumulative Transfers	-0.014^{**} [0.006]	-0.014^{**} [0.006]	-0.029^{***} [0.008]
Compensating transfer at 2003 median <i>Cumulative Transfers</i> (ten thousand pesos)	1.857** [0.859]	2.058* [1.099]	1.173*** [0.250]
Future Cumulative Transfers × 2003		0.006 [0.005]	-0.079^{***} [0.024]
Future Cumulative Transfers \times 2000n		-0.004^{***} [0.001]	-0.013*** [0.004]
Future Cumulative Transfers \times 2000m		0.001 [0.002]	-0.009^{***} [0.003]
Future Cumulative Transfers \times 1999		<0.001 [0.001]	-0.002 [0.002]
Future Cumulative Transfers × 1998		<0.001 [0.001]	> -0.001 [0.002]
Kleibergen-Paap Wald F-Stat on excluded variables			29
Observations R^2	30,414 0.105	30,414 0.105	30,414

TABLE 8—FUTURE TRANSFERS AND REFRIGERATOR ACQUISITION

Notes: All specifications include state by round-fixed effects and household controls described in the notes to Table 4. In column 3 excluded instruments are *PCT*, *Early* × *PCT*, and *Potential Future Cumulative Transfers* by round. Column 1 repeats results from Table 7. Robust standard errors clustered by village in brackets.

*** Significant at the 1 percent level.

**Significant at the 5 percent level.

*Significant at the 10 percent level.

Recertification.—Recertification could also make households forward looking. While households are promised payments for six more years after not being recertified as eligible, households may delay or avoid purchasing refrigerators to maintain future eligibility. This would bias the results of our main specification toward zero as current cumulative transfers are positively correlated with future transfers, and households with more at stake in the future would presumably be more reluctant to purchase a refrigerator to ensure continued participation in the program. The results in Table 8 are consistent with this. Controlling for future transfers, and thus the benefits of recertification, leads to bigger effects of transfers.

Self-Control.—We also considered the possibility that the difference in acquisition is driven by a lack of self-control. Particularly, the logic of the intertemporal optimization in Section II suggests that early households have the ability to replicate through saving the time path of transfers of the late households, but instead choose to allocate transfers differently. However, if these households lack self-control or access to savings vehicles, or are otherwise myopic it is possible that the temporal effects we observe are the consequence of households spending the transfers as they

receive them, rather than optimizing considering both current and future transfers. The negative coefficients on *Future Cumulative Transfers* in Table 8, however, are not consistent with lack of self-control or other myopic behavior.

Family Structure.—Another concern is that the nonlinear relationship between income and refrigerator acquisition reflects differences in both tastes and income across families with different structures. For example, the *Early* dummy is identified by considering households with the same cumulative income, some of whom received transfers steadily at low rates and some of whom received no transfers for 18 months and then high transfers once they began the program. Because of the randomization, the households in the early and late groups are similar on average. However by construction, households in the two groups with the same cumulative transfers have different family structures. The fact that specifications estimated with household fixed effects suggest larger effects than specifications that simply include household-level controls gives us reassurance that taste differences across households are not driving our results.

F. Placebo Tests

As an additional robustness check, we estimated cross-sectional placebo specifications using asset ownership before the households began receiving transfers. These specifications test whether the particular nonlinear relationship between family structure and transfers embodied in cumulative transfers predicts appliance ownership prior to the impact of the transfers themselves. The results are presented in Table 9. Each specification is estimated using instrumental variables and including household controls, comparable to the specification reported in column 4 of Table 7. Column 1 uses ownership from the baseline survey that was conducted in 1997 and cumulative transfers through 2003. The coefficients on *Cumulative Transfers, Early*, and *Early* × *Cumulative Transfers* are insignificantly different from zero.⁴⁴ This provides additional reassurance that the differential transfer rates experienced by households under Oportunidades are not systematically correlated with the propensity to acquire appliances.

Since the specifications at baseline in Table 9 are cross-sectional while the results in Table 7 were estimated as discrete time hazards, we estimated similar specifications for 2003 by way of comparison. These are presented in column 4. These specifications confirm the results in Table 4 and Table 7.

Columns 2 and 3 report another variant of the placebo test that predicts asset ownership in the period just before a household entered the program (baseline for treated households and 1999 for control households) as a function of the household's first-period transfers. These tests do not capture the variation induced by the randomization, but do use transfers in a period closer to the period in which households' baseline asset ownership is measured. If there were meaningful correlations between family structure (and thus transfer rates), such as the ages of children and girls versus boys and valuations of refrigerators, we should expect a relationship

⁴⁴Results are similar for placebo tests on other assets.

	Baseline ownership All households (1)	Baseline ownership Early households (2)	1999 ownership Late households (3)	2003 ownership All households (4)
Cumulative Transfers (2003)	0.006 [0.006]			0.054*** [0.012]
Early	0.005 [0.012]			0.029 [0.026]
Early \times Cumulative Transfers (2003)	-0.001 [0.005]			-0.030*** [0.011]
Cumulative Transfers (1998)		0.066 [0.151]		
Cumulative Transfers (2000m)			0.120 [0.224]	
Compensating transfer at 2003 median cumulative transfers (ten thousand pesos)	-0.675 [1.731]			1.686** [0.875]

TABLE 9—PLACEBO TEST: DO CUMULATIVE TRANSFERS PREDICT REFRIGERATOR OWNERSHIP AT BASELINE?

Notes: Columns 1–3 report results from three different placebo cross-sectional specifications, where the dependent variable is refrigerator ownership before transfers. For comparison, column 4 reports results using refrigerator ownership in 2003 as the dependent variable. All specifications include state-fixed effects and household controls, as described in the notes to Table 4. Columns 1 and 4, estimated using IV with *PCT* and *Early* × *PCT* as instruments, show no relationship between cumulative transfers at through 2003 and baseline ownership. Columns 2 and 3, estimated using IV with PCT as excluded instrument, show no statistical relationship between cumulative transfers in the first period available and ownership at the cross-section immediately prior to the beginning of transfers. Robust standard errors clustered by village in brackets. Kleibergen-Paap Wald *F*-stat on excluded variables not reported. All exceed 75.

***Significant at the 1 percent level.

**Significant at the 5 percent level.

*Significant at the 10 percent level.

between first-period transfer rates and asset ownership driven by preferences and not transfers. This test takes advantage of the fact that households with high first-period transfers are those with older and more female children, not necessarily those who will eventually have high cumulative transfers. Though noisier, these similarly show no statistically significant impact of first-period transfers. These tests also refute the hypothesis that household preferences for appliances change over time in a manner correlated with the transfer schedule, for example, if households with nine-year-old boys had similar preferences to households with nine-year-old girls, but preferences diverged when the children were in their teens.

VI. Conclusions

We develop a model and provide empirical analysis to demonstrate that households faced with credit constraints become much more likely to purchase energy-using assets with additional income once their income passes a threshold level. The timing of asset purchases in our data is also consistent with predictions from theory. Specifically, though households randomized to receive Oportunidades transfers earlier than other households faced greater expected benefits over the life of the program, they were less likely to acquire refrigerators. Given that refrigerators are long-lived assets, this is inconsistent with perfect capital markets and forward-looking consumers. Over the next several decades, wide-scale poverty alleviation programs as well as continued economic growth will lift the incomes of many of the world's poor. At the same time, governments, major multilateral and bilateral aid organizations, and private companies are spending hundreds of billions of dollars a year rolling out energy infrastructure in the developing world. As incomes rise and as energy access increases, families formerly living in poverty will for the first-time purchase refrigerators, water pumps, air conditioners, washing machines, vehicles, and other energy-using assets.

Our results have implications for both micro- and macro-level policies. Our results speak to the potential impacts of credit availability on asset acquisition. There is anecdotal evidence that retailers are beginning to sell appliances on credit in some parts of the developing world. For example, Wal-Mart's Banco de Wal-Mart recently began offering credit cards for purchases at Wal-Mart. This appliance-directed credit should lower the income levels at which households acquire appliances, but also could reduce the aggregate nonlinear relationship between income and energy use at a macro level. Our results also show that the time path of the payments in government transfer programs may impact asset acquisition rates. In the presence of credit constraints, the timing and frequency of transfer payments and the path of economic growth are important determinants of asset acquisition, not just transfer amounts.

As more and more households in the developing world become first-time owners of energy-using assets like refrigerators and vehicles, policies that promote more energy-efficient appliances, such as subsidies for high efficiency models, labeling or other information programs, or minimum efficiency standards, become increasingly important. While a considerable literature considers US households' willingness to trade off a high upfront cost for an energy efficient model versus lower usage costs, more work needs to be done to understand household energy efficiency decisions in the developing world, where the range of products offered, information about efficiency, energy prices, credit availability, and industrial organization of appliance sales all may differ.⁴⁵

At a more macro level, our results could help improve the accuracy of energy demand forecasts. While energy forecasters concur that the bulk of the growth in energy demand and associated greenhouse gas emissions is likely to come from the developing world (EIA 2013; International Energy Agency 2014), existing forecasts have not allowed for the possible nonlinear relationship between income growth, especially pro-poor growth, and energy demand suggested by our model (Wolfram, Shelef, and Gertler 2012). For example, we find evidence of this when we look historically at near-term forecasts and compare them to realized energy demand. For example, in its World Energy Outlook 2000, the Energy Information Administration forecast that China's energy demand in 2005 would be 55 quadrillion BTUs. Actual energy demand that year was a stunning 25 percent higher. In part, this reflects the fact that the EIA underestimated China's GDP growth, but even adjusting for

⁴⁵ Davis, Fuchs, and Gertler (2014) analyze the "Cash for Coolers" program, through which the Mexican government offered subsidies to households that exchanged older, presumably inefficient, refrigerators and air conditioners for new models. They show that the program led to significantly lower energy savings than ex ante analyses had forecast.

the higher GDP growth, the estimate is still 15 percent too low. Similarly, the EIA underestimated energy demand in Brazil, a country with aggressive antipoverty programs, for 2005 by more than 25 percent. By contrast, in India, a country that has had less success than Brazil and China lifting the income of the very poor, the EIA estimate for 2005 was much closer, only low by 3 percent.⁴⁶

Our results suggest that there will likely be a surge in the demand for energy as more and more of the households currently living in poverty benefit from overall economic development or explicit antipoverty programs and enter the middle class. Acquiring an energy-using asset for the first time leads to a considerable increase in a household's energy use. While income growth also affects energy consumption on the intensive margin, the effect is trivial compared to the effect of accumulating more energy-using assets.⁴⁷ Beyond ongoing use, acquisition of energy-using assets also induces indirect energy use through the manufacturing and distribution of the asset.⁴⁸ Investments in energy infrastructure require long lead times, so if the global demand for energy increases faster than anticipated, there could be significant shortages and increases in energy prices. In addition, country-specific energy forecasts are key inputs into projections of damages related to climate change as well as mitigation plans. Thus, it is critical to understand phenomenon like the one identified in this paper that will drive future growth in energy demand.

REFERENCES

- Aghion, Philippe, and Patrick Bolton. 1997. "A Theory of Trickle-Development." *Review of Economic Studies* 64 (2): 151–172.
- Albarran, Pedro, and Orazio Attanasio. 2004. "Do Public Transfers Crowd Out Private Transfers?: Evidence from a Randomized Experiment in Mexico." In *Insurance Against Poverty*, edited by Stefan Dercon, 281–304. Oxford: Oxford University Press.
- Allcott, Hunt, Allan Collard-Wexler, and Stephen D. O'Connell. 2016. "How Do Electricity Shortages Affect Industry? Evidence from India." *American Economic Review* 106 (3): 587–624.
- Attanasio, Orazio, Costas Meghir, and Ana Santiago. 2012. "Education Choices in Mexico: Using a Structural Model and a Randomized Experiment to Evaluate Progresa." *Review of Economic Studies* 79 (1): 37–66.
- Banerjee, Abhijit. 2004. "Contracting Constraints, Credit Markets, and Economic Development." In Advances in Economics and Econometrics: Theory and Applications, VIII World Congress of the Econometric Society, 3, edited by Mathias Dewatripoint, Lars P. Hansen, and Stephen J. Turnovsky, 1–46. New York: Cambridge University Press.
- Banerjee, Abhijit, and Andrew F. Newman. 1993. "Occupational Choice and the Process of Development." *Journal of Political Economy* 101 (2): 274–98.
- Barreca, Alan, Karen Clay, Olivier Deschenes, Michael Greenstone, and Joseph Shapiro. Forthcoming. "Adapting to Climate Change: The Remarkable Decline in the U.S. Temperature-Mortality Relationship over the 20th Century." *Journal of Political Economy*.
- Bento, Antonio, Larry Goulder, Mark Jacobsen, and Roger von Haefen. 2009. "Distributional and Efficiency Impacts of Increased US Gasoline Taxes." *American Economic Review* 99 (3): 667–99.
- Besley, Timothy, Stephen Coate, and Glenn Loury. 1993. "The Economics of Rotating Savings and Credit Associations." *American Economic Review* 83 (4): 792–810.

⁴⁷ In online Appendix E, we show that Oportunidades households' electricity consumption respond very little to transfers conditional on assets.
 ⁴⁸ Existing work in engineering estimates "life-cycle energy estimates" to include indirect energy consumption,

⁴⁶China, Brazil, and India are the only three countries in the developing world for which the Energy Information Administration provided country-specific forecasts.

⁴⁸Existing work in engineering estimates "life-cycle energy estimates" to include indirect energy consumption, such as the energy used to manufacture the refrigerator and the energy embodied in the steel. For a refrigerator, existing estimates suggest that the embodied energy is about ten percent of the lifetime energy (Gonzalez, Chase, and Horowitz 2012). Given that the lifetime is more than ten years, though, the embodied energy will more than double the energy consumption in the year it is purchased.

- Bonus, Holger. 1973. "Quasi-Engel Curves, Diffusion, and the Ownership of Major Consumer Durables." *Journal of Political Economy* 81 (3): 655–77.
- **Busse, Meghan, Christopher R. Knittel, and Florian Zettelmeyer.** 2013. "Are Consumers Myopic? Evidence from New and Used Car Purchases." *American Economic Review* 103 (1): 220–56.
- Chinese Household Income Project. 2002. ICPSR 21741. https://www.icpsr.umich.edu/icpsrweb/ICPSR/ studies/21741.
- Comisión Federal de Electricidad. 2012. "Meeting the Dual Goal of Energy Access and Sustainability: CSP Deployment in Mexico." http://www.esmap.org/sites/esmap.org/files/CFE_Meeting_dual_ goal_Mexico.pdf.
- Dargay, Joyce, Dermot Gately, and Martin Sommer. 2007. "Vehicle Ownership and Income Growth, Worldwide: 1960–2030." Energy Journal 28 (4): 143–170.
- **Davis, Lucas, Alan Fuchs, and Paul Gertler.** 2014. "Cash for Coolers: Evaluating a Large-Scale Appliance Replacement Program in Mexico." *American Economic Journal: Economic Policy* 6 (4): 207–38.
- Davis, Lucas, and Lutz Kilian. 2011. "The Allocative Cost of Price Ceilings in the U.S. Residential Market for Natural Gas." *Journal of Political Economy* 119 (2): 212–41.
- de Mel, Suresh, David McKenzie, and Christopher Woodruff. 2008. "Returns to Capital in Microenterprises: Evidence from a Field Experiment." *Quarterly Journal of Economics* 123 (4): 1329–72.
- **Demographic and Health Surveys (DHS).** *DHS Democratic and Health Survey for Kenya and Nigeria.* United States Agency for International Development: The DHS Program.
- **Dinkelman, Taryn.** 2011. "The Effects of Rural Electrification on Employment: New Evidence from South Africa." *American Economic Review* 101 (7): 3078–3108.
- **Dubin, Jeffrey A., and Daniel L. McFadden.** 1984. "An Econometric Analysis of Residential Electric Appliance Holdings and Consumption." *Econometrica* 52 (2): 345–62.
- Elbers, Chris, Jean O. Lanjouw, and Peter Lanjouw. 2003. "Micro-Level Estimation of Poverty and Inequality." *Econometrica* 71 (1): 355–64.
- Energy Information Administration (EIA). 2011. World Energy Projection System Plus: Residential Model. Washington, DC.
- Energy Information Administration (EIA). 2013. International Energy Outlook 2013. Washington, DC. http://www.eia.gov/forecasts/archive/ieo13/.
- Farrell, M. J. 1954. "The Demand for Motor-Cars in the United States." Journal of the Royal Statistical Society 117 (2): 171–201.
- Gertler, Paul, Sebastien Martinez, and Marta Rubio-Codina. 2012. "Investing Cash Transfers to Raise Long-Term Living Standards." American Economic Journal: Economic Policy 4 (1): 164–192.
- Gertler, Paul J., Orie Shelef, Catherine D. Wolfram, and Alan Fuchs. 2016. "The Demand for Energy-Using Assets among the World's Rising Middle Classes: Dataset." *American Economic Review*. http://dx.doi.org/10.1257/aer.20131455.
- Gonzalez, Amanda, Alex Chase, and Noah Horowitz. 2012. What We Know and Don't Know about Embodied Energy and Greenhouse Gases for Electronics, Appliances, and Light Bulbs. American Council for an Energy Efficient Economy Summer Study on Energy Efficiency in Buildings. http:// aceee.org/files/proceedings/2012/data/papers/0193-000301.pdf.
- Hoddinott, John, and Emmanuel Skoufias. 2004. "The Impact of Progress on Food Consumption." Economic Development and Cultural Change 53 (1): 37–61.
- **Hossian, Mahabub.** 1988. Credit for Alleviation of Rural Poverty: The Grameen Bank in Bangladesh. International Food Policy Research Institute.
- Instituto Brasileiro de Geografia e Estatistica. 2009. "National Household Sample Survey PNAD."
- Instituto Nacional de Estadistica, Geografia e Informatica, Mexico. 1992. "Encuesta Nacional de Ingresos y Gastos de los Hogares (ENIGH)."
- Instituto Nacional de Estadistica, Geografia e Informatica, Mexico. 1994–2011. "Usuarios de energía eléctrica."
- Instituto Nacional de Estadistica, Geografia e Informatica, Mexico. 1996. "Encuesta Nacional de Ingresos y Gastos de los Hogares (ENIGH)."
- Instituto Nacional de Estadistica, Geografia e Informatica, Mexico. 2004. "Encuesta Nacional de Ingresos y Gastos de los Hogares (ENIGH)."
- Instituto Nacional de Estadistica, Geografia e Informatica, Mexico. 2008. "Encuesta Nacional de Ingresos y Gastos de los Hogares (ENIGH)."
- International Energy Agency (IEA). 2014. World Energy Outlook, 2014. Paris, France: International Energy Agency.
- Johnson, Todd, Claudio Alatorre, Zayra Romo, and Feng Liu. 2009. Low-Carbon Development for Mexico. Washington, DC: The World Bank.

- Karlan, Dean, and Jonathan Morduch. 2010. "Access to Finance," In Handbook of Development Economics, Vol. 5, edited by Dani Rodrik and Mark Rosenzweig, 4703–84. Amsterdam: Elsevier Press.
- Kopits, Elizabeth, and Maureen L. Cropper. 2005. "Traffic Fatalities and Economic Growth." *Accident Analysis and Prevention* 37 (1): 169–178.
- Letschert, Virginie, and Michael McNeil. 2007. "Coping with Residential Electricity Demand in India's Future How Much Can Efficiency Achieve?" Lawrence Berkeley National Laboratory (LBNL 63199).
- Lindh, Thomas, and Henry Ohlsson. 1998. "Self-Employment and Wealth Inequality." *Review of Income and Wealth* 44 (1): 25–42.
- Lipscomb, Molly, A. Mushfiq Mobarak, and Tania Barham. 2013. "Development Effects of Electrification: Evidence from the Geologic Placement of Hydropower Plants in Brazil." *American Economic Journal: Applied Economics* 5 (2): 200–31.
- Lloyd-Ellis, Huw, and Dan Bernhardt. 2000. "Enterprise, Inequality and Economic Development." *Review of Economic Studies* 67 (1): 147–168.
- Ministry of Statistics and Programme Implementation. 2008. National Sample Survey. Government of India.
- National Bureau of Statistics. 2001. China Statistical Yearbook, 2001. Beijing, China. http://www.stats.gov.cn/english/statisticaldata/yearlydata/YB2001e/ml/indexE.htm.
- National Bureau of Statistics. 2011. China Statistical Yearbook, 2011. Beijing, China. http://www.stats.gov.cn/tjsj/ndsj/2011/indexeh.htm.
- Parker, Susan, and Emmanuel Skoufias. 2000. "The Impact of PROGRESA on Work, Leisure, and Time Allocation." International Food Policy Research Institute Discussion Paper. http://ebrary.ifpri. org/cdm/ref/collection/p15738coll2/id/125439 (accessed March 11, 2016).
- Parker, Susan, and Petra Todd. Forthcoming. "Conditional Cash Transfers: The Case of Progresa/ Oportunidades." Journal of Economic Literature.
- PR Newswire. 2014. "Refrigeration Industry Leaders Organize Global Food Cold Chain Council." New York, September 23. http://www.prnewswire.com/news-releases/refrigeration-industryleaders-organize-global-food-cold-chain-council-276595941.html (accessed March 11, 2016).
- Programa de Inclusion Social (PROSPERA). 1997. Encuesta de Características Socioeconomicas de los Hogares (ENCASEH).
- Programa de Inclusion Social (PROSPERA). 1998, 1999, 2000, 2003, 2007. Households Evaluation Survey (ENCEL).
- **RAND Corporation.** 2004. National Socio-Economic Household Survey (SUSENAS), Indonesia. Santa Monica, CA.
- Rapson, David. 2014. "Durable Goods and Long-Run Electricity Demand: Evidence from Air Conditioner Purchase Behavior." Journal of Environmental Economics and Management 68 (1): 141–60.
- Rosenzweig, Mark, and Kenneth Wolpin. 1993. "Credit Market Constraints, Consumption Smoothing, and the Accumulation of Durable Production Assets in Low-Income Countries: Investment in Bullocks in India." *Journal of Political Economy* 101 (2): 223–44.
- Schiraldi, Pasquale. 2011. "Automobile Replacement: A Dynamic Structural Approach." RAND Journal of Economics 42 (2): 266–91.
- Schultz, T. Paul. 2004. "School Subsidies for the Poor: Evaluating the Mexican Progress Poverty Program." *Journal of Development Economics* 74 (1): 199–250.
- Skoufias, Emmanuel, and Susan Parker. 2001. "Conditional Cash Transfers and Their Impact on Child Work and Schooling: Evidence from the PROGRESA Program in Mexico." *Economía* 2 (1): 45–86.
- State Statistical Bureau of the People's Republic of China. 2001, 2011. China Statistical Yearbook. Beijing.
- Twilley, Nicola. 2014. "What Do Chinese Dumplings Have to Do With Global Warming?" New York Times, July 25.
- United States Census Bureau. 2011. Extended Measures of Well-Being: Living Conditions in the United States: 2011. Washington, DC. https://www.census.gov/prod/2013pubs/p70-136.pdf.
- Wolfe, Barbara L., and Jere R. Behrman. 1982. "Determinants of Child Mortality, Health, and Nutrition in a Developing Country." *Journal of Development Economics* 11 (2): 163–193.
- Wolfram, Catherine, Orie Shelef, and Paul Gertler. 2012. "How Will Energy Demand Develop in the Developing World?" *Journal of Economic Perspectives* 26 (1): 119–138.
- World Bank. 2008. The Welfare Impact of Rural Electrification: A Reassessment of the Costs and Benefits. Washington, DC.
- World Bank. 2011. World Bank World Development Indicators. Washington, DC.
- Zhou, Nan, David Fridely, Michael McNeil, Nina Zheng, Virginie Letschert, Jing Ke, and Yarmina Saheb. 2011. "Analysis of Potential Energy Saving and CO₂ Emission Reduction of Home Appliances and Commercial Equipment in China." *Energy Policy* 39 (8): 4541–50.

This article has been cited by:

- 1. Teresa Randazzo, Filippo Pavanello, Enrica De Cian. 2023. Adaptation to climate change: Airconditioning and the role of remittances. *Journal of Environmental Economics and Management* 120, 102818. [Crossref]
- 2. Meera Mahadevan, Robyn Meeks, Takashi Yamano. 2023. Reducing information barriers to solar adoption: Experimental evidence from India. *Energy Economics* **120**, 106600. [Crossref]
- 3. Dambala Gelo, Umakrishnan Kollamparambil, Marc Jeuland. 2023. The causal effect of income on household energy transition: Evidence from old age pension eligibility in South Africa. *Energy Economics* **119**, 106538. [Crossref]
- 4. Averi Chakrabarti, Sudhanshu Handa. 2023. The impacts of cash transfers on household energy choices. *American Journal of Agricultural Economics* **3**. . [Crossref]
- 5. # #. 2023. Household Energy Consumption Transition: The Impact of the Internet Use and Off-Farm Employment. *Sustainable Development* 13:01, 172-185. [Crossref]
- Lucie Maruejols, Lisa Höschle, Xiaohua Yu. 2022. Vietnam between economic growth and ethnic divergence: A LASSO examination of income-mediated energy consumption. *Energy Economics* 114, 106222. [Crossref]
- 7. Seema Jayachandran. 2022. How Economic Development Influences the Environment. *Annual Review of Economics* 14:1, 229-252. [Crossref]
- 8. Matthew E. Kahn, Somik Lall. Will the Developing World's Growing Middle Class Support Low-Carbon Policies? 1, . [Crossref]
- 9. Mathilde Lebrand. Rising Incomes, Transport Demand, and Sector Decarbonization 135, . [Crossref]
- Jianglong Li, Mun Sing Ho, Chunping Xie, Nicholas Stern. 2022. China's flexibility challenge in achieving carbon neutrality by 2060. *Renewable and Sustainable Energy Reviews* 158, 112112. [Crossref]
- 11. Justin Caron, Thibault Fally. 2022. Per Capita Income, Consumption Patterns, and CO 2 Emissions. *Journal of the Association of Environmental and Resource Economists* **9**:2, 235-271. [Crossref]
- 12. Akshaya Jha, Louis Preonas, Fiona Burlig. 2022. Blackouts in the Developing World: The Role of Wholesale Electricity Markets. SSRN Electronic Journal 91. . [Crossref]
- 13. Mathilde Sylvie Maria Lebrand, Ewane Theophile. 2022. Rising Incomes, Transport Demand, and Sector Decarbonization. *SSRN Electronic Journal* 45. . [Crossref]
- 14. Sean Fox, Felix Agyemang, rashid memon. 2022. Mapping Urban Living Standards and Economic Activity in Developing Countries with Energy Data. *SSRN Electronic Journal* **3399**. [Crossref]
- 15. Matthew E. Kahn, Somik Lall. 2022. Will the Developing World's Growing Middle Class Support Low Carbon Policies?. SSRN Electronic Journal 31. . [Crossref]
- 16. Filippo Pavanello, Enrica De Cian, Marinella Davide, Malcolm Mistry, Talita Cruz, Paula Bezerra, Dattakiran Jagu, Sebastian Renner, Roberto Schaeffer, André F. P. Lucena. 2021. Air-conditioning and the adaptation cooling deficit in emerging economies. *Nature Communications* 12:1. [Crossref]
- 17. Maria Alice Moz-Christofoletti, Paula Carvalho Pereda. 2021. Distributional welfare and emission effects of energy tax policies in Brazil. *Energy Economics* **104**, 105616. [Crossref]
- Fydess Khundi-Mkomba, Akshay Kumar Saha, Umaru Garba Wali. 2021. Examining the state of energy poverty in Rwanda: An inter-indicator analysis. *Heliyon* 7:11, e08441. [Crossref]
- Cristian Huse, Claudio Lucinda, Andre Ribeiro. 2021. Assessing the effects of a large temporary energy savings program: Evidence from a developing country. *Journal of Environmental Economics and Management* 110, 102529. [Crossref]

- Shaun D. McRae, Frank A. Wolak. 2021. Retail pricing in Colombia to support the efficient deployment of distributed generation and electric stoves. *Journal of Environmental Economics and Management* 110, 102541. [Crossref]
- Jing Jin, Yucheng Wang, Xinye Zheng. 2021. District heating versus self-heating: Estimation of energy efficiency gap using regression discontinuity design. *China Economic Quarterly International* 1:3, 208-220. [Crossref]
- 22. A P Behrer, R J Park, G Wagner, C M Golja, D W Keith. 2021. Heat has larger impacts on labor in poorer areas *. *Environmental Research Communications* 3:9, 095001. [Crossref]
- 23. Lucas Davis, Paul Gertler, Stephen Jarvis, Catherine Wolfram. 2021. Air conditioning and global inequality. *Global Environmental Change* 69, 102299. [Crossref]
- 24. Miguel Poblete-Cazenave, Shonali Pachauri. 2021. A model of energy poverty and access: Estimating household electricity demand and appliance ownership. *Energy Economics* **98**, 105266. [Crossref]
- 25. Radhika Khosla, Anna Agarwal, Neelanjan Sircar, Deepaboli Chatterjee. 2021. The what, why, and how of changing cooling energy consumption in India's urban households. *Environmental Research Letters* 16:4, 044035. [Crossref]
- 26. Nasir Iqbal, Saima Nawaz. 2021. Cash transfers and residential demand for electricity: insights from BISP, Pakistan. *Environmental Science and Pollution Research* 28:12, 14401-14422. [Crossref]
- 27. Jin Guo, Chuanzhong Li, Chu Wei. 2021. Decoupling economic and energy growth: aspiration or reality?. *Environmental Research Letters*. [Crossref]
- 28. Gopal K. Sarangi, Farhad Taghizadeh-Hesary. Market-Led Energy Efficiency Transformation in India: A Deep Dive into the Perform, Achieve, Trade (PAT) Scheme 223-242. [Crossref]
- 29. Esther Acquah. 2021. Labor Supply and Welfare Effects of Electricity in Ghana: Does Geography Matter?. SSRN Electronic Journal 52. . [Crossref]
- 30. Seema Jayachandran. 2021. How Economic Development Influences the Environment. SSRN Electronic Journal 1. . [Crossref]
- 31. Anthony Amoah, Kofi Korle. 2020. Forest depletion in Ghana: the empirical evidence and associated driver intensities. *Forestry Economics Review* **2**:1, 61-80. [Crossref]
- Shalu Agrawal, S.P. Harish, Aseem Mahajan, Daniel Thomas, Johannes Urpelainen. 2020. Influence of improved supply on household electricity consumption - Evidence from rural India. *Energy* 211, 118544. [Crossref]
- Muhammad Bilal Tahir, Muhammad Sohaib, Muhammad Rafique, Muhammad Sagir, Najeeb Ur Rehman, Shabbir Muhammad. 2020. Visible light responsive photocatalytic hydrogen evolution using MoS2 incorporated ZnO. *Applied Nanoscience* 10:10, 3925-3931. [Crossref]
- 34. Setu Pelz, Johannes Urpelainen. 2020. Measuring and explaining household access to electrical energy services: Evidence from rural northern India. *Energy Policy* 145, 111782. [Crossref]
- Srishti Gupta, Eshita Gupta, Gopal K. Sarangi. 2020. Household Energy Poverty Index for India: An analysis of inter-state differences. *Energy Policy* 144, 111592. [Crossref]
- 36. Danyi Qi, John W. Apolzan, Ran Li, Brian E. Roe. 2020. Unpacking the decline in food waste measured in Chinese households from 1991 to 2009. *Resources, Conservation and Recycling* 160, 104893. [Crossref]
- 37. Alberto Salvo. 2020. Local Pollution as a Determinant of Residential Electricity Demand. *Journal of the Association of Environmental and Resource Economists* **7**:5, 837-872. [Crossref]
- Marco Baudino. 2020. Environmental Engel curves in Italy: A spatial econometric investigation. *Papers in Regional Science* 99:4, 999-1018. [Crossref]

- 39. Maximiliane Sievert, Jevgenijs Steinbuks. 2020. Willingness to pay for electricity access in extreme poverty: Evidence from sub-Saharan Africa. *World Development* **128**, 104859. [Crossref]
- 40. Lucas W. Davis, Sebastian Martinez, Bibiana Taboada. 2020. How effective is energy-efficient housing? Evidence from a field trial in Mexico. *Journal of Development Economics* 143, 102390. [Crossref]
- 41. Daniele Malerba. 2020. Poverty alleviation and local environmental degradation: An empirical analysis in Colombia. *World Development* **127**, 104776. [Crossref]
- 42. Cristian Huse, Claudio Lucinda, Andre Ribeiro Cardoso. 2020. Consumer response to energy label policies: Evidence from the Brazilian energy label program. *Energy Policy* **138**, 111207. [Crossref]
- 43. Leontina Pinto, Jacques Szczupak, Robinson Semolini. Load Forecast by Multi-Task Learning Models: Designed for a New Collaborative World 389-399. [Crossref]
- 44. Léopold T. Biardeau, Lucas W. Davis, Paul Gertler, Catherine Wolfram. 2020. Heat exposure and global air conditioning. *Nature Sustainability* **3**:1, 25-28. [Crossref]
- 45. Jennifer Richmond, Johannes Urpelainen. 2019. Electrification and appliance ownership over time: Evidence from rural India. *Energy Policy* **133**, 110862. [Crossref]
- 46. Juan-Pablo Carvallo, Jay Taneja, Duncan Callaway, Daniel M. Kammen. 2019. Distributed Resources Shift Paradigms on Power System Design, Planning, and Operation: An Application of the GAP Model. *Proceedings of the IEEE* 107:9, 1906-1922. [Crossref]
- 47. Yating Li, Yinxin Fei, Xiao-Bing Zhang, Ping Qin. 2019. Household appliance ownership and income inequality: Evidence from micro data in China. *China Economic Review* 56, 101309. [Crossref]
- Hongmei Yi, Di Mo, Huan Wang, Qiufeng Gao, Yaojiang Shi, Paiou Wu, Cody Abbey, Scott Rozelle. 2019. Do Resources Matter? Effects of an In-Class Library Project on Student Independent Reading Habits in Primary Schools in Rural China. *Reading Research Quarterly* 54:3, 383-411. [Crossref]
- 49. Mingzhi Hu, Jiaqi Liu, Xue Wang. Consumption Propensity and Entrepreneurship: Evidence from China 265-286. [Crossref]
- 50. Daniele Malerba. 2019. Poverty-energy-emissions pathways: Recent trends and future sustainable development goals. *Energy for Sustainable Development* **49**, 109-124. [Crossref]
- 51. Arik Levinson, James O'Brien. 2019. Environmental Engel Curves: Indirect Emissions of Common Air Pollutants. *The Review of Economics and Statistics* 101:1, 121-133. [Crossref]
- 52. Alberto Salvo. 2018. Electrical appliances moderate households' water demand response to heat. *Nature Communications* 9:1. . [Crossref]
- 53. Mohammed Bouznit, María Pablo-Romero, Antonio Sánchez-Braza. 2018. Residential Electricity Consumption and Economic Growth in Algeria. *Energies* 11:7, 1656. [Crossref]
- 54. Daire McCoy, John Curtis. 2018. Exploring the spatial and temporal determinants of gas central heating adoption. *Resource and Energy Economics* 52, 64-86. [Crossref]
- 55. Antoine Levesque, Robert C. Pietzcker, Lavinia Baumstark, Simon De Stercke, Arnulf Grübler, Gunnar Luderer. 2018. How much energy will buildings consume in 2100? A global perspective within a scenario framework. *Energy* 148, 514-527. [Crossref]
- 56. Joshua Goodman, Michael Hurwitz, Jisung Park, Jonathan Smith. 2018. Heat and Learning. SSRN Electronic Journal 91. . [Crossref]
- 57. Branko Bošković, Ujjayant Chakravorty, Martino Pelli, Anna Risch. 2018. The Effect of Forest Access on the Market for Fuelwood in India. *SSRN Electronic Journal*. [Crossref]
- 58. Geoffrey Barrows, Teevrat Garg, Akshaya Jha. 2018. The Economic Benefits versus Environmental Costs of India's Coal Fired Power Plants. *SSRN Electronic Journal*. [Crossref]

- 59. Jacopo Bonan, Stefano Pareglio, Massimo Tavoni. 2017. Access to modern energy: a review of barriers, drivers and impacts. *Environment and Development Economics* 22:5, 491-516. [Crossref]
- 60. Rohan Best, Paul Burke. 2017. The Importance of Government Effectiveness for Transitions toward Greater Electrification in Developing Countries. *Energies* 10:9, 1247. [Crossref]
- 61. Marianne Fay, Luis Alberto Andrés, Charles Fox, Ulf Narloch, Stéphane Straub, Michael Slawson. What Lies Ahead for the Region's Infrastructure? 55-88. [Crossref]
- 62. María del P. Pablo-Romero, Rafael Pozo-Barajas, Rocío Yñiguez. 2017. Global changes in residential energy consumption. *Energy Policy* **101**, 342-352. [Crossref]
- 63. María del P. Pablo-Romero, Rafael Pozo-Barajas, Antonio Sánchez-Braza. 2016. Analyzing the effects of Energy Action Plans on electricity consumption in Covenant of Mayors signatory municipalities in Andalusia. *Energy Policy* **99**, 12-26. [Crossref]
- 64. Jacopo Bonan, Stefano Pareglio, Massimo Tavoni. 2016. Access to Modern Energy: A Review of Barriers, Drivers and Impacts. SSRN Electronic Journal. [Crossref]